## **RESOLUTION NO. 2249**

A RESOLUTION OF THE CITY COUNCIL OF THE CITY OF GOLDEN AUTHORIZING THE EXECUTION OF A PURCHASE CONTRACT WITH IRON KEY HOMES LLC FOR THE CITY OWNED LOT AT 826 JOSEPH CIRCLE

WHEREAS, City Council on December 13, 2012 adopted Ordinance No. 1925 declaring as surplus the City owned lot at 826 Joseph Circle; and

WHEREAS, Iron Key Homes LLC has submitted a purchase offer consistent with the conditions of Ordinance No. 1925, and

WHEREAS, City Council wishes authorize the contract.

THEREFORE, BE IT RESOLVED BY THE CITY COUNCIL OF THE CITY OF GOLDEN, COLORADO:

Section 1: The Contract to Buy and Sell Real Estate is approved in substantially the same form as the copy attached hereto as Exhibit A and made a part of this resolution. The Mayor is authorized to execute the agreement and such documents as are necessary to effectuate the conveyance on behalf of the City of Golden, subject to the review and approval by the City Attorney.

Adopted this  $28^{th}$  day of February, 2013.

Marjorle N. Sloan

Mayor

Susan M. Brooks, MMC

City Clerk

APPROVED AS TO FORM:

David S. Williamson

City Attorney

I, Susan M. Brooks, City Clerk of the City of Golden, Colorado, do hereby certify that the foregoing is a true copy of a certain Resolution adopted by the City Council of the City of Golden, Colorado at a rescheduled regular business meeting thereof held on the  $28^{th}$  day of February, 2013.

ATTEST:

Susan M. Brooks, City Clerk of the City of

Golden, Colorado

# Colorado Land Professionals, LLC

Douglas Ferraro

Ph: 303.888.7436 Fax: 303.273.5694

The printed portions of this form, except differentiated additions, have been approved by the Colorado Real Estate Commission. (CBS4-9-12) (Mandatory 1-13)

THIS FORM HAS IMPORTANT LEGAL CONSEQUENCES AND THE PARTIES SHOULD CONSULT LEGAL AND TAX OR OTHER COUNSEL BEFORE SIGNING.

# CONTRACT TO BUY AND SELL REAL ESTATE (LAND)

( Property with No Residences)  $(\Box$  Property with Residences-Residential Addendum Attached)

|  | Date: 2/19/2013                                   |
|--|---|
| AGREEMENT  |   |
| 1. AGREEMENT. Buyer, identified in § 2.1, agrees to buy, and Seller, identified below on the terms and conditions set forth in this contract (   | entified in § 2.3, agrees to sell, the Contract). |
| 2. PARTIES AND PROPERTY.   |   |
| 2.1. Buyer. Buyer. Iron Key Homes, LLC , will take title to the Prope  | rty described below as D Joint                    |
| Tenants In Common Other Limited Liability Company  2.2. Assignability and Inurement. This Contract Shall Shall without Seller's prior written consent. Except as so restricted, this Contract shall binding upon the heirs, personal representatives, successors and assigns of the  2.3. Seller. Seller City of Golden, is the current owner of the Prope  2.4. Property. The Property is the following legally described real estat  | inure to the benefit of and be parties.           |
| Colorado:  | e in the County of                                |
| LOT 1, BLOCK 2, FILING NO. 2, CANYON VIEW SUBDIVISION  |   |
| known as No. 826 Joesph Circle Golden CO 80401.  |   |
| together with the interests, easements, rights, benefits, improvements and attack and all interest of Seller in vacated streets and alleys adjacent thereto, except as 2.5. Inclusions. The Purchase Price includes the following items (Included 2.5.1. Fixtures. All fixtures attached to the Property on the date of the Property on the date of the Property of of th | s herein excluded (Property).                     |
| Other Fixtures: NONE   |   |
| If any fixtures are attached to the Property after the date of this Contract, such as in the Purchase Price.  2.5.2. Personal Property. If on the Property whether attached of the following items are included:   |   |
| the following items are included: NONE   |   |
| Other Personal Property: NONE  | *   |
| The Personal Property to be conveyed at Closing shall be conveyed taxes (except personal property taxes for the year of Closing), liens and encumbring NONE  Conveyance shall be by bill of sale or ot   | rances, except                                    |
| 2.5.3. Trade Fixtures. With respect to trade fixtures, Seller and I  | Buyer agree as follows:                           |
| The Trade Fixtures to be conveyed at Closing shall be conveyed by S (except personal property taxes for the year of Closing). Iiens and encumbrances. <b>NONE</b>  | Seller free and clear of all taxes<br>except      |
| Conveyance shall be by bill of sale or other applicable legal instrument.  2.5.4. Water Rights, Water and Sewer Taps.  | *   |
| NONE  2.5.4.1. Deeded Water Rights. The following legally described to the second state of the second stat | ribed water rights:                               |
| Any water rights shall be conveyed by \(\sigma\) \(n/a\) instrument.   | eed Other applicable legal                        |
| 2.5.4.2. Well Rights. If any water well is to be transferred   | ed to Buyer, Seller agrees to                     |
| DNTRACT TO BUY AND SELL REAL ESTATE - Land Page 1 of 15  |   |
| rage 1 01 15   |   |
| Seller(c) Initials   |   |

CBS4-9-12. CC

Buyer(s) Initials

supply required information about such well to Buyer. Buyer understands that if the well to be transferred is a Small Capacity Well or a Domestic Exempt Water Well used for ordinary household purposes, Buyer shall, prior to or at Closing, complete a Change in Ownership form for the well. If an existing well has not been registered with the Colorado Division of Water Resources in the Department of Natural Resources (Division), Buyer shall complete a registration of existing well form for the well and pay the cost of registration. If no person will be providing a closing service in connection with the transaction, Buyer shall file the form with the Division within sixty days after Closing. The Well Permit # is N/A

NONE

2.5.4.3. Water Stock Certificates:

2.5.4.4. Sewer Tap

Note: Buyer is advised to obtain, from the provider, written confirmation of the amount remaining to be paid, if any, time and other restrictions for transfer and use of the tap.

2.5.4.5. Other Rights:

NONE

NONE

NONE

2.5.5. Growing Crops. With respect to growing crops, Seller and Buyer agree as follows:

**2.6. Exclusions.** The following items are excluded (Exclusions):

DATES AND DEADLINES.

| Item No. | Reference | Event  | Date or Deadline   |
|----------|-----------|--|--|
| 1        | § 4.3     | Alternative Earnest Money Deadline                 | MARCH 5TH, 2013  |
|          |           | Title and Association                              |  |
| 2        | § 7.1     | Record Title Deadline                              | MARCH 6TH. 2013  |
| 3        | § 7.5     | Exceptions Request Deadline                        | MARCH 6TH, 2013  |
| 4        | § 8.1     | Record Title Objection Deadline                    | MARCH 21ST, 2013   |
| 5        | § 8.2     | Off-Record Title Deadline                          | MARCH 6TH, 2013  |
| 6        | § 8.2     | Off-Record Title Objection Deadline                | MARCH 21ST, 2013   |
| 7        | § 8.3     | Title Resolution Deadline                          | MARCH 26TH, 2013   |
| 8        | § 7.6     | Association Documents Deadline                     | MARCH 6TH, 2013  |
| 9        | § 7.6     | Association Documents Objection Deadline           | MARCH 21ST, 2013   |
| 10       | § 8.5     | Right of First Refusal Deadline                    | N/A  |
|          | 5         | Seller's Property Disclosure                       |  |
| 11       | § 10.1    | Seller's Property Disclosure Deadline              | NA   |
|          | L         | oan and Credit                                     |  |
| 12       | § 5.1     | Loan Application Deadline                          | N/A  |
| 13       | § 5.2     | Loan Objection Deadline                            | NA   |
| 14       | § 5.3     | Buyer's Credit Information Deadline                | NA   |
| 5        | § 5.3     | Disapproval of Buyer's Credit Information Deadline | N/A  |
| 6        | § 5.4     | Existing Loan Documents Deadline                   | N/A  |
| 7        | § 5.4     | Existing Loan Documents Objection Deadline         | N/A  |
| 8        | § 5.4     | Loan Transfer Approval Deadline                    | N/A  |
|          | Δ.        | ppraisal   | филороння в в проводения проводения по том по проводения в нев филосопроводения по   |
| 9        | § 6.2     | Appraisal Deadline                                 | N/A  |
| 0        | § 6.2     | Appraisal Objection Deadline                       | N/A  |
|          | S         | urvey  |  |
| 1        | § 9.1     | Current Survey Deadline                            | N/A  |
| 2        | § 9.2     | Current Survey Objection Deadline                  | N/A  |
|          | Ir        | nspection and Due Diligence                        | A CONTRACTOR OF THE PROPERTY O |
| 3        | § 10.2    | Inspection Objection Deadline                      | APRIL 7th, 2013  |
| 4        | § 10.3    | Inspection Resolution Deadline                     | APRIL 11TH, 2013   |
|          | § 10.5    | Property Insurance Objection Deadline              | N/A  |
|          | § 10.6    | Due Diligence Documents Delivery Deadline          | MARCH STH 2012   |

|                | 24     | 9 10.3       | Inspection Resolution Deadline            | APRIL 11TH, 2013 |
|----------------|--------|--------------|---|------------------|
|                | 25     | § 10.5       | Property Insurance Objection Deadline     | N/A              |
|                | 26     | § 10.6       | Due Diligence Documents Delivery Deadline | MARCH 5TH, 2013  |
|                |        |              |   |                  |
| CBS4-9-12.     | CONTRA | CT TO BUY AN | ID SELL REAL ESTATE - Land Page 2 of 15   |                  |
|                |        | A Home       |   |                  |
| Buyer(s) Initi | als    | //           | Seller(s) Initials                        |                  |

| 27 | § 10.7 | Due Diligence Documents Objection Deadline    | APRIL 1ST, 2013                  |
|----|--------|---|----------------------------------|
| 28 | § 10.8 | Environmental Inspection Objection Deadline   | APRIL 11TH, 2013                 |
| 29 | § 10.8 | ADA Evaluation Objection Deadline             | N/A                              |
| 30 | § 11.1 | Tenant Estoppel Statements Deadline           | N/A                              |
| 31 | § 11.2 | Tenant Estoppel Statements Objection Deadline | N/A                              |
|    |        | Closing and Possession                        |                                  |
| 32 | § 12.3 | Closing Date                                  | APRIL 12TH, 2013                 |
| 33 | § 17   | Possession Date                               | DOD                              |
| 34 | § 17   | Possession Time                               | IMMEDIATLEY<br>FOLLOWING CLOSING |
| 35 | § 28   | Acceptance Deadline Date                      | MARCH 1ST, 2013                  |
| 36 | § 28   | Acceptance Deadline Time                      | 5:00 P.M                         |
| 37 | n/a    | n/a   | n/a                              |
| 38 | n/a    | n/a   | n/a                              |

Note: Applicability of Terms.

179 180

187

188

189 190

191 192 193

194

195 196 197

198

199 200

201

223

224 225

226

227 228

231

234

235

236 237

238

239

240 241

242

243

246

247

Any box, blank or line in this Contract left blank or completed with the abbreviation "N/A", or the word "Deleted" means such provision in Dates and Deadlines (§ 3), including any deadline, is not applicable and the corresponding provision of this Contract to which reference is made is deleted.

The abbreviation "MEC" (mutual execution of this Contract) means the date upon which both parties have signed this Contract.

#### PURCHASE PRICE AND TERMS.

4.1. Price and Terms. The Purchase Price set forth below shall be payable in U.S. Dollars by Buyer as follows:

| Item No. | Reference | Item                        | Amount       | Amount       |
|----------|-----------|-----------------------------|--------------|--------------|
| 1        | § 4.1     | Purchase Price              | \$100,000.00 |              |
| 2        | § 4.2     | Earnest Money               |              | \$5,000.00   |
| 3        | § 4.5     | New Loan                    |              |              |
| 4        | § 4.6     | Assumption Balance          |              |              |
| 5        | § 4.7     | Seller or Private Financing |              |              |
| 6        | n/a       | n/a                         |              |              |
| 7        | n/a       | n/a                         |              |              |
| 8        | § 4.3     | Cash at Closing             |              | \$95,000.00  |
| 9        |           | TOTAL                       | \$100,000.00 | \$100,000.00 |

4.2. Seller Concession. Seller, at Closing, shall credit, as directed by Buyer, an amount of to assist with any and all of the following: Buyer's closing costs, (Seller Concession). Seller Concession is in addition to any sum Seller has agreed to pay or credit Buyer elsewhere in this Contract. Seller Concession will be reduced to the extent it exceeds the aggregate of what is allowed by Buyer's lender as set forth in the Closing Statement, Closing Disclosure or HUD-1, at Closing.

Earnest Money. The Earnest Money set forth in this section, in the form of 4.3. **COMPANY CHECK** , shall be payable to and held by TITLE COMPANY

(Earnest Money Holder), in its trust account, on behalf of both Seller and Buyer. The Earnest Money deposit shall be tendered, by Buyer, with this Contract unless the parties mutually agree to an Alternative Earnest Money Deadline (§ 3) for its payment. The parties authorize delivery of the Earnest Money deposit to the company conducting the Closing (Closing Company), if any, at or before Closing. In the event Earnest Money Holder has agreed to have interest on Earnest Money deposits transferred to a fund established for the purpose of providing affordable housing to Colorado residents, Seller and Buyer acknowledge and agree that any interest accruing on the Earnest Money deposited with the Earnest Money Holder in this transaction shall be transferred to such fund.

4.3.1. Alternative Earnest Money Deadline. The deadline for delivering the Earnest Money, if other than at the time of tender of this Contract is as set forth as the Alternative Earnest Money Deadline (§ 3).

4.3.2. Return of Earnest Money. If Buyer has a Right to Terminate and timely terminates, Buyer shall be entitled to the return of Earnest Money as provided in this Contract. If this Contract is terminated as set forth in § 25 and, except as provided in § 24, if the Earnest Money has not already been returned following receipt of a Notice to Terminate, Seller agrees to execute and return to Buyer or Broker working with Buyer. written mutual instructions, i.e., Earnest Money Release form, within three days of Seller's receipt of such form.

Page 3 of 15

Form of Funds; Time of Payment; Funds Available. CBS4-9-12. CONTRACT TO BUY AND SELL REAL ESTATE - Land

| Buyer(s) Initials | Seller(s) Initials |
|-------------------|--------------------|
|-------------------|--------------------|

| 249<br>250<br>251<br>252<br>253                      | proceeds, Cash at Closing and closing costs, shall be in funds that comply with all applicable Colorado laws, including electronic transfer funds, certified check, savings and loan teller's check and cashier's check (Good Funds).  |
|--|--|
| 254<br>258<br>256<br>257<br>258<br>258<br>259<br>260 | 4.4.2. Available Funds. All funds required to be paid at Closing or as otherwise agreed in writing between the parties shall be timely paid to allow disbursement by Closing Company at Closing OR SUCH PARTY SHALL BE IN DEFAULT. Buyer represents that Buyer, as of the date of this Contract, ☑ Does Not have funds that are immediately verifiable and available in an amount not less than the amount stated as   |
| 261<br>262   | 4.5. New Loan. (Omitted as inapplicable)   |
| 263<br>264<br>265                                    | <b>4.6.</b> Assumption. (Omitted as inapplicable)  |
| 266<br>267<br>268                                    | <b>4.7. Seller or Private Financing.</b> (Omitted as inapplicable)   |
| 269<br>270   |  |
| 271<br>272   |  |
| 273<br>274<br>275                                    | 5. FINANCING CONDITIONS AND OBLIGATIONS. (Omitted as inapplicable)   |
| 276<br>277<br>278                                    | 5.3. Credit Information and Buyer's New Senior Loan. (Omitted as inapplicable)   |
| 279<br>280<br>281                                    | 5.4. Existing Loan Review. (Omitted as inapplicable)   |
| 282<br>283   | 6 APPRAISAL PROVISIONS   |
| 284  | 6.1. Lender Property Requirements. If the lender imposes any requirements or repairs (Requirements)  |
| 285<br>286   | to the transfer of the transfe |
| 287<br>288   | Contract, Seller has the Right to Terminate under § 25.1, (notwithstanding § 10 of this Contract), on or before three  |
| 289  | subjective discretion. Seller's Right to Terminate in this 6.6.1 shall not apply if on or before any termination by  |
| 290<br>291   | Soller pursuant to this S. G. 1: (1) the parties enter into a written assessment reporting the Description and (2) the   |
| 292  | Requirements have been completed; or (3) the satisfaction of the Requirements is waived in writing by Buyer.   |
| 293<br>294   | Tippi aloui obligation. The applicable Appliabal provision set forth below shall apply to the  |
| 295  | respective loan type set forth in § 4.5.3, or if a cash transaction, i.e. no financing, § 6.2.1 shall apply.   |
| 296<br>297   | <b>6.2.1. Conventional/Other.</b> Buyer has the sole option and election to terminate this Contract if the Property's valuation, determined by an appraiser engaged on behalf of   |
| 298  | the Property's valuation, determined by an appraiser engaged on behalf of is less than the Purchase Price. The appraisal shall be received by Buyer or Buyer's lender on or before <b>Appraisal Deadline</b>   |
| 299<br>300   | (§ 3). Buyer has the Right to Terminate under § 25.1, on or before <b>Appraisal Objection Deadline</b> (§ 3), if the   |
| 301<br>302   | Property's valuation is less than the Purchase Price and Seller's receipt of either a copy of such appraisal or  |
| 303  | written notice from lender that confirms the Property's valuation is less than the Purchase Price. This § 6.2.1 is for   |
| 304<br>305   | the sole benefit of Buyer.   |
| 306  | 6.3. Cost of Appraisal. Cost of any appraisal to be obtained after the date of this Contract shall be  |
| 307<br>308   | timely paid by Buyer Seller. The cost of the appraisal may include any and all fees paid to the appraiser, appraisal management company, lender's agent or all three.  |
| 309  | appraisa management company, lender a agent of all tillee.   |
| 310<br>311   | 7. EVIDENCE OF TITLE AND ASSOCIATION DOCUMENTS.  |
| 312<br>313   | 7.1. Seller Selects Title Insurance Company If this box is checked, Seller shall select the title  |
| 314  | insurance company to furnish the owner's title insurance policy at Seller's expense. On or before Record Title   |
| 315<br>316   | Deadline (§ 3), Seller shall furnish to Buyer a current commitment for owner's title insurance policy (Title   |
| 317  | Commitment), in an amount equal to the Purchase Price, or if this box is checked an Abstract of title certified  |
| 318<br>319   | to a current date. Seller shall cause the title insurance policy to be issued and delivered to Buyer as soon as practicable at or after Closing.   |
| 320  | 7.2. Buyer Selects Title Insurance Company. If this box is checked, Buyer shall select the   |
| 321<br>322   | title insurance company to furnish the owner's title insurance policy at Buyer's expense. On or before <b>Record</b>   |
| 323  | Title Deadline (§ 3), Buyer shall furnish to Seller, a current commitment for owner's title insurance policy (Title  |
| 324<br>325   | Commitment), in an amount equal to the Purchase Price.   |
| 326<br>327   | If neither box in § 7.1 or § 7.2 is checked, § 7.1 applies.  |
| 328  | 7.3. Owner's Extended Coverage (OEC). The Title Commitment ☑ Shall ☐ Shall Not commit to delete or insure over the standard exceptions which relate to: (1) parties in possession, (2) unrecorded  |
| 329<br>330   | easements, (3) survey matters, (4) unrecorded mechanics' liens, (5) gap period (effective date of commitment to  |
| 331  | date deed is recorded), and (6) unpaid taxes, assessments and unredeemed tax sales prior to the year of Closing  |
| CBS4-9-12.   | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 4 of 15   |
| Buyer(s) Init  | ials   |
|  | Schol (3) Illitals   |

| 33:           |  |  |
|---------------|--|--|
| 334           | 4 <b>Note:</b> The title insurance company may not agree to delete or insure over any or all of the standard exceptions.   |  |
| 33!<br>33!    | Tions Tromain for Ozom the title insurance company agrees to provide air endorsement for   |  |
| 33            | DEC, any additional premium expense to obtain an endorsement for OEC shall be paid by Buyer - Seller   |  |
| 338           |  |  |
| 339<br>340    | 7.7. Dayer's hight to neview this commitment and this bocuments. Buyer has the right to  |  |
| 34:           | review the Title Commitment, its provisions and Title Documents (defined in § 7.5), and if not satisfactory to Buyer,  |  |
| 342<br>343    | ,  |  |
| 344           | Company) As Copies of Exceptions. Others the box in § 7.2 is checked (buyer Selects fille insurance  |  |
| 345<br>346    |  |  |
| 347           | declarations, covenants, conditions and restrictions burdening the Property, and (2) if a Title Commitment is  |  |
| 348<br>349    |  |  |
| 350           |  |  |
| 351<br>352    | College shall have the obligation to furnish these decuments appeared to this postion if requested by Duney and time   |  |
| 353           |  |  |
| 354           | record in the office of the clerk and recorder in the county where the Property is located. The Abstract or Title  |  |
| 355<br>356    | Commitment together with any conjector summeries of such decuments furnished pursuant to this section  |  |
| 357           | constitute the title documents (collectively, Title Documents).  |  |
| 358<br>359    | The Existing About dots of Title. Oction shall deliver to buyer dopies of any abstracts of title   |  |
| 360           | covering an or any portion of the Property (Abstract) in Seller's possession on or before <b>Record Title Deadline</b> (§  |  |
| 361           |  |  |
| 362<br>363    | Tremes Association Postments. Homeowich Association Postments (Association   |  |
| 364           | Documents) consist of the following:   |  |
| 365<br>366    |  |  |
| 367           | 7.6.2 Migutes of most recent annual experts' mosting   |  |
| 368<br>369    |  |  |
| 370           |  |  |
| 371<br>372    | (88.7.6.1, 7.6.2 and 7.6.3 collectively Governing Decuments)   |  |
| 373           |  |  |
| 374           | sheet, (2) annual and most recent income and expenditures statement. (3) annual budget, and (4) reserve study, if  |  |
| 375<br>376    | any (conectively, Financial Documents).  |  |
| 377           | 7.6.5. Common Interest Community Disclosure. THE PROPERTY IS LOCATED WITHIN A  |  |
| 378<br>379    | The state of the s |  |
| 380           | OWNER OF THE PROPERTY WILL BE REQUIRED TO BE A MEMBER OF THE OWNER'S ASSOCIATION FOR   |  |
| 381<br>382    |  |  |
| 383           |  |  |
| 384<br>385    | A COMPONENT INC. OF THE LOCAL COLUMN ASSETS ASSETT ASSETS ASSETS ASSETS ASSETS ASSETS ASSETS ASSETS ASSETT ASSETS ASSETS ASSETS ASSETT ASSETS ASSETT  |  |
| 386           | ASSOCIATION COULD PLACE A LIEN ON THE PROPERTY AND POSSIBLY SELL IT TO PAY THE DEBT.   |  |
| 387<br>388    | THE DECLARATION, BYLAWS, AND RULES AND REGULATIONS OF THE COMMUNITY MAY PROHIBIT THE   |  |
| 389           | OWNER FROM MAKING CHANGES TO THE PROPERTY WITHOUT AN ARCHITECTURAL REVIEW BY THE   |  |
| 390           | ASSOCIATION (OR A COMMITTEE OF THE ASSOCIATION) AND THE APPROVAL OF THE ASSOCIATION.   |  |
| 391<br>392    | PURCHASERS OF PROPERTY WITHIN THE COMMON INTEREST COMMUNITY SHOULD INVESTIGATE THE   |  |
| 393           | FINANCIAL OBLIGATIONS OF MEMBERS OF THE ASSOCIATION. PURCHASERS SHOULD CAREFULLY   |  |
| 394<br>395    | READ THE DECLARATION FOR THE COMMUNITY AND THE BYLAWS AND RULES AND REGULATIONS OF   |  |
| 396           | THE ASSOCIATION.   |  |
| 397<br>398    | 7.6.6. Association Documents to Buyer.   |  |
| 399           | - Color to Floride Association Documents. Seller Shall cause the Association   |  |
| 400<br>401    | Documents to be provided to Buyer, at Seller's expense, on or before <b>Association Documents Deadline</b> (§ 3).  7.6.6.2. Seller Authorizes Association. Seller authorizes the Association to provide the  |  |
| 402           | Association Documents to Buyer, at Seller's expense.   |  |
| 403           | 7.6.6.3. Seller's Obligation. Seller's obligation to provide the Association Documents   |  |
| 404<br>405    | shall be fulfilled upon Buyer's receipt of the Association Documents, regardless of who provides such documents.   |  |
| 406           | Note: If neither box in this § 7.6.6 is checked, the provisions of § 7.6.6.1 shall apply.  |  |
| 407<br>408    | 7.6.7. Conditional on Buyer's Review. Buyer has the right to review the Association  |  |
| 409           | Documents. Buyer has the Right to Terminate under § 25.1, on or before Association Documents Objection   |  |
| 410<br>411    | Deadline (§ 3), based on any unsatisfactory provision in any of the Association Documents, in Buyer's sole   |  |
| 412           | subjective discretion. Should Buyer receive the Association Documents after Association Documents Deadline   |  |
| 413           | (§ 3), Buyer, at Buyer's option, has the Right to Terminate under § 25.1 by Buyer's Notice to Terminate received   |  |
| CBS4-9-12.    | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 5 of 15   |  |
|               |  |  |
| Buyer(s) Init | ials Seller(s) Initials  |  |

by Seller on or before ten days after Buyer's receipt of the Association Documents. If Buyer does not receive the Association Documents, or if Buyer's Notice to Terminate would otherwise be required to be received by Seller 417 after Closing Date (§ 3), Buyer's Notice to Terminate shall be received by Seller on or before Closing (§ 12.3). If 418 Seller does not receive Buyer's Notice to Terminate within such time, Buyer accepts the provisions of the 419 420 421 422 423 424 425 426 427 428 429 430 431 433 434 435 436 437 438 439 440 441 442 443 Documents as satisfactory. 444 445 446 448 449 450 451 452 453 454 455 456 457 458 459 460 461 463 464 465 466 467

468

469 470

471

472 473

474

475

476 477

478

479

480

481 482

483

486

487

488 489

490

491

492

493 494

495

496 497

# Association Documents as satisfactory, and Buyer waives any Right to Terminate under this provision, notwithstanding the provisions of § 8.5.

RECORD TITLE AND OFF-RECORD TITLE. 8.1. Record Title. Buyer has the right to review and object to any of the Title Documents (Right to Object to Title, Resolution) as set forth in § 8.3. Buyer's objection may be based on any unsatisfactory form or content of Title Commitment, notwithstanding § 13, or any other unsatisfactory title condition, in Buyer's sole subjective discretion. If Buyer objects to any of the Title Documents, Buyer shall cause Seller to receive Buyer's Notice to Terminate or Notice of Title Objection on or before Record Title Objection Deadline (§ 3). If Title Documents are not received by Buyer, on or before the Record Title Deadline (§ 3), or if there is an endorsement to the Title Commitment that adds a new Exception to title, a copy of the new Exception to title and the modified Title Commitment shall be delivered to Buyer. Buyer shall cause Seller to receive Buyer's Notice to Terminate or Notice of Title Objection on or before ten days after receipt by Buyer of the following documents: (1) any required Title Document not timely received by Buyer, (2) any change to the Title Documents, or (3) endorsement to the Title Commitment. If Seller receives Buyer's Notice to Terminate or Notice of Title Objection, pursuant to this § 8.1 (Record Title), any title objection by Buyer and this Contract shall be governed by the provisions set forth in § 8.3 (Right to Object to Title, Resolution). If Seller does not receive Buyer's Notice to Terminate or Notice of Title Objection by the applicable deadline specified above, Buyer accepts the condition of title as disclosed by the Title

- 8.2. Off—Record Title. Seller shall deliver to Buyer, on or before Off—Record Title Deadline (§ 3), true copies of all existing surveys in Seller's possession pertaining to the Property and shall disclose to Buyer all easements, liens (including, without limitation, governmental improvements approved, but not yet installed) or other title matters (including, without limitation, rights of first refusal and options) not shown by public records, of which Seller has actual knowledge (Off-Record Matters). Buyer has the right to inspect the Property to investigate if any third party has any right in the Property not shown by public records (such as an unrecorded easement, unrecorded lease, boundary line discrepancy or water rights). Buyer's Notice to Terminate or Notice of Title Objection of any unsatisfactory condition (whether disclosed by Seller or revealed by such inspection, notwithstanding § 13), in Buyer's sole subjective discretion, shall be received by Seller on or before Off-Record Title Objection Deadline (§ 3). If Seller receives Buyer's Notice to Terminate or Notice of Title Objection pursuant to this § 8.2 (Off-Record Title), any title objection by Buyer and this Contract shall be governed by the provisions set forth in § 8.3 (Right to Object to Title, Resolution). If Seller does not receive Buyer's Notice to Terminate or Notice of Title Objection on or before Off-Record Title Objection Deadline (§ 3), Buyer accepts title subject to such rights, if any, of third parties of which Buyer has actual knowledge. Unless disclosed in writing, Seller represents and warrants that there are no Off-Record Matters.
- 8.3. Right to Object to Title, Resolution. Buyer's Right to Object to Title shall include, but not be limited to those matters set forth in §§ 8.1 (Record Title), 8.2 (Off-Record Title) and 13 (Transfer of Title), in Buyer's sole subjective discretion (collectively, Right to Object to Title). If Buyer objects to any title matter, on or before the applicable deadline, Buyer shall have the option to either (1) object to the condition of title, or (2) terminate this Contract.
- 8.3.1. Title Resolution. If Seller receives Buver's Notice of Title Objection, as provided in § 8.1 (Record Title) or § 8.2 (Off-Record Title), on or before the applicable deadline, and if Buyer and Seller have not agreed to a written settlement thereof on or before Title Resolution Deadline (§ 3), this Contract shall terminate on the expiration of Title Resolution Deadline (§ 3) unless Seller receives Buyer's written withdrawal of Buyer's Notice of Title Objection, (i.e., Buyer's written notice to waive objection to such items and waives the right to terminate for that reason), on or before expiration of Title Resolution Deadline (§ 3).
- **8.3.2.** Right to Terminate Title Objection. Buyer has the Right to Terminate under § 25.1, on or before the applicable deadline, based on any unsatisfactory title matter, in Buyer's sole subjective discretion.
- 8.4. Special Taxing Districts. SPECIAL TAXING DISTRICTS MAY BE SUBJECT TO GENERAL OBLIGATION INDEBTEDNESS THAT IS PAID BY REVENUES PRODUCED FROM ANNUAL TAX LEVIES ON THE TAXABLE PROPERTY WITHIN SUCH DISTRICTS. PROPERTY OWNERS IN SUCH DISTRICTS MAY BE PLACED AT RISK FOR INCREASED MILL LEVIES AND TAX TO SUPPORT THE SERVICING OF SUCH DEBT WHERE CIRCUMSTANCES ARISE RESULTING IN THE INABILITY OF SUCH A DISTRICT TO DISCHARGE SUCH INDEBTEDNESS WITHOUT SUCH AN INCREASE IN MILL LEVIES. BUYERS SHOULD INVESTIGATE THE SPECIAL TAXING DISTRICTS IN WHICH THE PROPERTY IS LOCATED BY CONTACTING THE COUNTY TREASURER, BY REVIEWING THE CERTIFICATE OF TAXES DUE FOR THE PROPERTY, AND BY OBTAINING FURTHER INFORMATION FROM THE BOARD OF COUNTY COMMISSIONERS, THE COUNTY CLERK AND RECORDER, OR THE COUNTY ASSESSOR.

Buyer has the Right to Terminate under § 25.1, on or before Off-Record Title Objection Deadline (§ 3), based on any unsatisfactory effect of the Property being located within a special taxing district, in Buyer's sole

Page 6 of 15

|                   | 1 - |                    |
|-------------------|-----|--------------------|
| Buyer(s) Initials | THE | Seller(s) Initials |

CONTRACT TO BUY AND SELL REAL ESTATE - Land

subjective discretion. 8.5. Right of First Refusal or Contract Approval. If there is a right of first refusal on the Property, or a right to approve this Contract, Seller shall promptly submit this Contract according to the terms and conditions of such right. If the holder of the right of first refusal exercises such right or the holder of a right to approve disapproves this Contract, this Contract shall terminate. If the right of first refusal is waived explicitly or expires, or the Contract is approved, this Contract shall remain in full force and effect. Seller shall promptly notify Buyer in writing of the foregoing. If expiration or waiver of the right of first refusal or Contract approval has not occurred on or before Right of First Refusal Deadline (§ 3), this Contract shall then terminate. 8.6. Title Advisory. The Title Documents affect the title, ownership and use of the Property and should be reviewed carefully. Additionally, other matters not reflected in the Title Documents may affect the title, ownership and use of the Property, including, without limitation, boundary lines and encroachments, area, zoning, unrecorded easements and claims of easements, leases and other unrecorded agreements, and various laws and governmental regulations concerning land use, development and environmental matters. The surface estate may be owned separately from the underlying mineral estate, and transfer of the surface estate does not necessarily include transfer of the mineral rights or water rights. Third parties may hold interests in oil, gas, other minerals, geothermal energy or water on or under the Property, which interests may give them rights to enter and use the Property. Such matters may be excluded from or not covered by the title insurance policy. Buyer is advised to timely consult legal counsel with respect to all such matters as there are strict time limits provided in this Contract [e.g., Record Title Objection Deadline (§ 3) and Off-Record Objection Deadline (§ 3)]. CURRENT SURVEY REVIEW. 9.1. Current Survey Conditions. If the box in § 9.1.1 or § 9.1.2 is checked, Buyer, the issuer of the Title Commitment or the provider of the opinion of title if an Abstract, and \_\_\_\_ shall receive a Current Survey, i.e., Improvement Location Certificate, Improvement Survey Plat or other form of survey set forth in § 9.1.2 (collectively, Current Survey), on or before Current Survey Deadline (§ 3). The Current Survey shall be certified by the surveyor to all those who are to receive the Current Survey. 9.1.1. Improvement Location Certificate. If the box in this § 9.1.1 is checked, □Seller ☑Buyer shall order or provide, and pay, on or before Closing, the cost of an Improvement Location Certificate. 9.1.2. Other Survey. If the box in this § 9.1.2 is checked, a Current Survey, other than an Improvement Location Certificate, shall be an Improvement Survey Plat I parties agree that payment of the cost of the Current Survey and obligation to order or provide the Current Survey shall be as follows: 9.2. Current Objection. Buyer has the right to review and object to the Current Survey. Buyer has the Right to Terminate under § 25.1, on or before the Current Survey Objection Deadline (§ 3), if the Current Survey is not timely received by Buyer or based on any unsatisfactory matter with the Current Survey, notwithstanding § 8.2 or § 13. DISCLOSURE, INSPECTION AND DUE DILIGENCE PROPERTY DISCLOSURE, INSPECTION, INDEMNITY, INSURABILITY, DUE DILIGENCE AND 10. SOURCE OF WATER. 10.1. Seller's Property Disclosure. On or before Seller's Property Disclosure Deadline (§ 3), Seller agrees to deliver to Buyer the most current version of the applicable Colorado Real Estate Commission's Seller's Property Disclosure form completed by Seller to Seller's actual knowledge, current as of the date of this Contract, 10.2. Inspection Objection. Unless otherwise provided in this Contract, Buyer acknowledges that Seller is conveying the Property to Buyer in an "as is" condition, "where is" and "with all faults". Seller shall disclose to Buyer, in writing, any latent defects actually known by Seller. Buyer, acting in good faith, has the right to have inspections (by one or more third parties, personally or both) of the Property and Inclusions (Inspection), at Buyer's expense. If (1) the physical condition of the Property, including but not limited to, the roof, walls, structural integrity of the Property, the electrical, plumbing, HVAC and other mechanical systems of the Property, (2) the physical condition of the Inclusions, (3) service to the Property (including utilities and communication services), systems and components of the Property, e.g. heating and plumbing, (4) any proposed or existing transportation project, road, street or highway, or (5) any other activity, odor or noise (whether on or off the Property) and its effect or expected effect on the Property or its occupants is unsatisfactory, in Buyer's sole subjective discretion, Buyer shall, on or before Inspection Objection Deadline (§ 3): 10.2.1. Notice to Terminate. Notify Seller in writing that this Contract is terminated; or 10.2.2. Inspection Objection. Deliver to Seller a written description of any unsatisfactory physical condition that Buyer requires Seller to correct. Buyer has the Right to Terminate under § 25.1, on or before Inspection Objection Deadline (§ 3) if the Property or Inclusions are unsatisfactory, in Buyer's sole subjective discretion.

499

500 501

502

504 505

506

508

509

511

512

513 514

515

516

519

520

521 522

523

524 525 526

527

528

529 530

531

533

534

535

538

539 540

542

543 544

545

546

547 548 549

550 551 552

553

554 555

556

557

558 559

560

562 563

564

565

566 567

568

570

572

573

574 575

576

577

578

CBS4-9-12. CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 7 of 15

Buyer(s) Initials Seller(s) Initials

- 10.3. Inspection Resolution. If an Inspection Objection is received by Seller, on or before Inspection Objection Deadline (§ 3), and if Buyer and Seller have not agreed in writing to a settlement thereof on or before Inspection Resolution Deadline (§ 3), this Contract shall terminate on Inspection Resolution Deadline (§ 3), unless Seller receives Buyer's written withdrawal of the Inspection Objection before such termination, i.e., on or before expiration of Inspection Resolution Deadline (§ 3).
- 10.4. Damage, Liens and Indemnity. Buyer, except as otherwise provided in this Contract or other written agreement between the parties, is responsible for payment for all inspections, tests, surveys, engineering reports, or other reports performed at Buyer's request (Work) and shall pay for any damage that occurs to the Property and Inclusions as a result of such Work. Buyer shall not permit claims or liens of any kind against the Property for Work performed on the Property at Buyer's request. Buyer agrees to indemnify, protect and hold Seller harmless from and against any liability, damage, cost or expense incurred by Seller and caused by any such Work, claim, or lien. This indemnity includes Seller's right to recover all costs and expenses incurred by Seller to defend against any such liability, damage, cost or expense, or to enforce this section, including Seller's reasonable attorney fees, legal fees and expenses. The provisions of this section shall survive the termination of this Contract. This § 10.4 does not apply to items performed pursuant to an Inspection Resolution.
- 10.5. Insurability. Buyer has the right to review and object to the availability, terms and conditions of and premium for property insurance (Property Insurance). Buyer has the Right to Terminate under § 25.1, on or before **Property Insurance Objection Deadline** (§ 3), based on any unsatisfactory provision of the Property Insurance, in Buyer's sole subjective discretion.
- 10.6. Due Diligence Documents. Seller agrees to deliver copies of the following documents and information pertaining to the Property (Due Diligence Documents) to Buyer on or before **Due Diligence** Documents Delivery Deadline(§ 3):
  - 10.6.1. All contracts relating to the operation, maintenance and management of the Property:
  - **10.6.2.** Property tax bills for the last <u>1</u> years;
- **10.6.3.** As-built construction plans to the Property and the tenant improvements, including architectural, electrical, mechanical, and structural systems; engineering reports; and permanent Certificates of Occupancy, to the extent now available;
  - 10.6.4. A list of all Inclusions to be conveyed to Buyer;
  - **10.6.5.** Operating statements for the past **0** years;
  - 10.6.6. A rent roll accurate and correct to the date of this Contract;
- **10.6.7.** All current leases, including any amendments or other occupancy agreements, pertaining to the Property (Leases);
- **10.6.8.** A schedule of any tenant improvement work Seller is obligated to complete but has not yet completed and capital improvement work either scheduled or in process on the date of this Contract;
- **10.6.9.** All insurance policies pertaining to the Property and copies of any claims which have been made for the past <u>0</u> years;
- **10.6.10.** Soils reports, Surveys and engineering reports or data pertaining to the Property (if not delivered earlier under § 8.2);
- **10.6.11.** Any and all existing documentation and reports regarding Phase I and II environmental reports, letters, test results, advisories, and similar documents respective to the existence or nonexistence of asbestos, PCB transformers, or other toxic hazardous or contaminated substances, and/or underground storage tanks and/or radon gas. If no reports are in Seller's possession or known to Seller, Seller warrants that no such reports are in Seller's possession or known to Seller;
- **10.6.12.** Any *Americans with Disabilities Act* reports, studies or surveys concerning the compliance of the Property with said Act;
- **10.6.13.** All permits, licenses and other building or use authorizations issued by any governmental authority with jurisdiction over the Property and written notice of any violation of any such permits, licenses or use authorizations, if any; and
  - 10.6.14. Other Documents:

#### NONE

- 10.7. Due Diligence Documents Conditions. Buyer has the right to review and object to Due Diligence Documents, zoning and any use restrictions imposed by any governmental agency with jurisdiction over the Property (Zoning), in Buyer's sole subjective discretion, and has the right to object if Seller fails to deliver to Buyer all Due Diligence Documents. Buyer shall also have the unilateral right to waive any condition herein.
- 10.7.1. Due Diligence Documents Objection. Buyer has the Right to Terminate under § 25.1, on or before **Due Diligence Documents Objection Deadline** (§ 3), based on any unsatisfactory matter with the Due Diligence Documents, in Buyer's sole subjective discretion. If all Due Diligence Documents under § 10.6 are not received by Buyer on or before **Due Diligence Documents Delivery Deadline** (§ 3), then Buyer has the Right to Terminate under § 25.1 on or before the earlier of ten days after **Due Diligence Documents Objection Deadline** (§ 3) or Closing.
- **10.7.2. Zoning.** Buyer has the Right to Terminate under § 25.1, on or before **Due Diligence Documents Objection Deadline** (§ 3), based on any unsatisfactory zoning, in Buyer's sole subjective discretion.

| CBS4-9-12.   | CONTRACT TO BUY AND SEI | LL REAL ESTATE - Land           | Page 8 of 15   |
|--------------|-------------------------|---------------------------------|----------------|
| Buyer(s) Ini | cials                   | Seller(s) In                    | nitials        |
|              |                         | CTMeContracts.com - ©2013 CTM S | Software Corp. |

| 66<br>66      | 10.7.3. Source of Potable Water (Residential Land and Residential Improvements Only).  |
|---------------|--|
| 66°           | Addendum displacing the source of notable water for the Down to Down Down Down Down Down Down Down Dow   |
| 669           | receipt of a copy of the current well permit.  |
| 670<br>671    | Note to Division COME WATER PROVIDED DELY TO VARIABLE PROPERTY ON MONREMENTARY E CROWN   |
| 673<br>673    |  |
| 674           | DETERMINE THE LONG-TERM SUFFICIENCY OF THE PROVIDER'S WATER SUPPLIES   |
| 67!<br>678    | 10.8. Due Diligence — Environmental, ADA. Buyer shall have the right to obtain environmental   |
| 673<br>678    |  |
| 679<br>680    | Environmental Site Assessment. Li Priase II  |
| 681           | Assessments) and/or $\square$ N/A , at the expense of $\square$ Seller $\square$ Buyer (Environmental Inspection). In  |
| 682<br>683    | addition, Buyer may also conduct an evaluation whether the Property compiles with the Americans with   |
| 684<br>685    |  |
| 686<br>687    | if any   |
| 688           | If Buyer's Phase I Environmental Site Assessment recommends a Phase II Environmental Site  |
| 689<br>690    | Assessment, the Environmental inspection objection beautifie (8 3) shall be extended by a day's (Extended  |
| 691<br>692    | Environmental inspection Objection Deadline) and it such Extended Environmental inspection Objection Deadline  |
| 693           | Buyer shall have the Right to Terminate under 8.25.1, on or before <b>Environmental Inspection Objection</b>   |
| 694<br>695    | Deadline (§ 3), or if applicable the Extended Environmental Inspection Objection Deadline, based on any  |
| 696<br>697    |  |
| 698<br>699    | trootief et an ADA Evaluation in Devote Ada Evaluation objection beautifie (§ 5), based on any   |
| 700           | 10.9. Existing Leases; Modification of Existing Leases; New Leases. Seller states that none of   |
| 701<br>702    | and because to be accigned to the buyer at the time of closing contain any tent concessions, tent reductions of  |
| 703<br>704    |  |
| 705           | without the prior written consent of Buyer, which consent shall not be unreasonably withheld or delayed  |
| 706<br>707    |  |
| 708<br>709    |  |
| 710<br>711    | Enterpol Statements College Holling and deliverate Described in Figure 1 Statement Statements  |
| 712           | Deadline (§ 3), statements in a form and substance reasonably acceptable to Buyer, from each occupant or   |
| 713<br>714    | omandments (Local) station   |
| 715<br>716    |  |
| 717<br>718    | 11.1.2. That said Lease is in full force and effect and that there have been no subsequent   |
| 719<br>720    | modifications of amendments;   |
| 721           | Seller;  |
| 722<br>723    | 11.1.4. The amount of monthly (or other applicable period) rental paid to Seller;  |
| 724<br>725    | 11.1.5. That there is no default under the terms of said Lease by landlord or occupant; and  |
| 726           |  |
| 727<br>728    | 11.2. Tenant Estoppel Statements Objection Buyer has the Right to Terminate under § 25.1, on or  |
| 729<br>730    | before Tenant Estoppel Statements Objection Deadline (§ 3), based on any unsatisfactory Estoppel   |
| 731<br>732    | Statement, in Buyer's sole subjective discretion or if Seller fails to deliver the Estoppel Statements on or before<br>Tenant Estoppel Statements Deadline (§ 3). Buyer shall also have the unilateral right to waive any          |
| 733           | unsatisfactory Estoppel Statement.   |
| 734<br>735    |  |
| 736<br>737    | CLOSING PROVISIONS   |
| 738<br>739    | 12. CLOSING DOCUMENTS, INSTRUCTIONS AND CLOSING.   |
| 740           | <ul> <li>12. CLOSING DOCUMENTS, INSTRUCTIONS AND CLOSING.</li> <li>12.1. Closing Documents and Closing Information. Seller and Buyer shall cooperate with the Closing</li> </ul>   |
| 741<br>742    | Company to enable the Closing Company to prepare and deliver documents required for Closing to Buyer and   |
| 743<br>744    | Seller and their designees. If Buyer is obtaining a new loan to purchase the Property, Buyer acknowledges  |
| 745<br>746    | Buyer's lender shall be required to provide the Closing Company in a timely manner all required loan documents and financial information concerning Buyer's new loan. Buyer and Seller will furnish any additional information and |
| 740           | documents required by Closing Company that will be necessary to complete this transaction. Buyer and Seller  |
| CBS4-9-12.    | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 9 of 15   |
|               |  |
| Buyer(s) Init | ials Seller(s) Initials  |

 $\label{lem:com-com-com} CTMe Contracts.com - @2013 \ CTM \ Software \ Corp.$ 

| Buyer(s) Init     | ials Seller(s) Initials  |
|-------------------|--|
|                   | rage 10 01 15  |
| 830<br>CBS4-9-12. | 16. PRORATIONS. The following shall be prorated to Closing Date (§ 3), except as otherwise provided:  CONTRACT TO BUY AND SELL REAL ESTATE - Land  Page 10 of 15   |
| 827<br>828<br>829 | shall be paid when due by 🛮 Buyer 🗆 Seller 🗀 One-Half by Buyer and One-Half by Seller 🗀 None.  |
| 824<br>825<br>826 | 15.7. Sales and Use Tax. Any sales and use tax that may accrue because of this transaction   |
| 822<br>823        | \$ <u>n/a</u> for $\square$ Augmentation Membership $\square$ Small Domestic Water Company $\square$ <u>n/a</u> and must be paid at Closing by $\square$ Buyer $\square$ Seller $\square$ One-Half by Buyer and One-Half by Seller $\square$ None.   |
| 820<br>821        | \$ n/a for \( \subseteq \text{Water Stock/ Certificates} \) Water District   |
| 818<br>819        | Contract, do not exceed:   |
| 816<br>817        | amount of <u>TBD</u> % of the Purchase Price or \$  15.6. Water Transfer Fees. The Water Transfer Fees can change. The fees, as of the date of this  |
| 814<br>815        | whether one or more, is for the following association(s): CANNON VIEW HOA in the total   |
| 813               | Closing by Buyer Seller One-Half by Buyer and One-Half by Seller None. The Private Transfer fee,   |
| 811<br>812        | 13.3. Private Transfer Fee. Private transfer fees and other fees due to a transfer of the Property,  |
| 809<br>810        | 15.5 Private Transfer Fee Private transfer fees and other fees due to a transfer of the Property   |
| 807<br>808        | 15.4. Local Transfer Tax.   The Local Transfer Tax of WA % of the Purchase Price   |
| 805<br>806        |  |
| 804               | Offe-half by Seller —None. Any record change fee assessed by the Association including, but not limited to,  |
| 802<br>803        | statement of assessments (Status Letter) shall be paid by Buyer Seller One-Half by Buyer and   |
| 800<br>801        | 15.2 Status Lotter and Decord Change Face. Any face incident to the incurred of Association's  |
| 798<br>799        | □ Other N/A  |
| 796<br>797        | 15.2. Closing Services Fee. The fee for real estate closing services shall be paid at Closing by   |
| 795               | all other items required to be paid at Closing, except as otherwise provided herein.   |
| 793<br>794        | 15.1 Closing Costs Developed Advantage in Costs Indiana in the impact in |
| 791<br>792        |  |
| 789<br>790        | Closing from the proceeds of this transaction or from any other course   |
| 787<br>788        | 14. PAYMENT OF ENCUMBRANCES. Any encumbrance required to be paid shall be paid at or before  |
| 785<br>786        | io.s. Chief with the seneratin country olerk and necessaer's office  |
| 784               | of Canyon View per document recorded at Reception No. 2007030761,  |
| 782<br>783        | the associated Designations of Covenants Conditions and Postvictions   |
| 780<br>781        |  |
| 778<br>779        | Current Survey Review (3 9),   |
| 777               | Buyer has actual knowledge and which were accepted by Buyer in accordance with Off-Record Title (§ 8.2) and  |
| 775               | 10.2. Distribution unity easements (including cable 19),   |
| 773<br>774        | , , ,  |
| 771<br>772        | 13.1. Those specific Exceptions described by reference to recorded documents as reflected in the   |
| 770               | hereon, whether assessed or not. Title shall be conveyed subject to:   |
| 768<br>769        | the general taxes for the year of Closing. Except as provided nerein, title shall be conveyed free and clear of all  |
| 768<br>763        | accept and clear of an taxes except  |
| 76!               | Buyer with the other terms and provisions hereof, Seller shall execute and deliver a good and sufficient   |
| 763<br>764        | 13 TRANSEER OF TITLE Subject to tender of payment at Closing as required berein and compliance by  |
| 76:<br>76:        | companies).  |
| 759<br>760        | 2011100 vary bottlesh american between providers (c.g., atterneys, ichaers, inspectors and title   |
| 75°<br>758        | 12.4. Disclosure of Settlement Costs. Buyer and Seller acknowledge that costs, quality, and extent of  |
| 75                | abell he as designated by THE DATTEC   |
| 75-<br>75         | date specified as the Closing Date (§ 3) or by mutual agreement at an earlier date. The hour and place of Closing  |
| 75:<br>75:        | 12.3 Clasing Dolivory of dood from Coller to Puyer shall be at aloging (Clasing). Clasing shall be an the  |
| 75<br>75          |  |
| 74                | shall sight and complete all customary or reasonably required documents at or before Closing.  |

CTMeContracts.com - ©2013 CTM Software Corp.

| 832<br>833       |  |
|------------------|--|
| 834              | Closing  |
| 835<br>836       | wost necent will Levy and most necent Assessed valuation, or - Other   |
| 837<br>838       | n/a  |
| 839              | tropofor or gradit to Buyor the approximation of the last and the last |
| 840<br>841       |  |
| 842              | Buyer all Leases in effect at Closing and Buyer shall assume Seller's obligations under such Leases.   |
| 843<br>844       | 16.2 According Accomments Current regular According accomments and duce (According   |
| 845              | Assessments) paid in advance shall be credited to Seller at Closing. Cash reserves held out of the regular   |
| 846<br>847       | resolution resolution of actioned maintenance by the resolution shall not be decided to concretate as  |
| 848              | may be otherwise provided by the Governing Documents. Buyer acknowledges that Buyer may be obligated to pay  |
| 849              |  |
| 850<br>851       | bloshing bate (3 0) by the Association shall be the obligation of Library  |
| 852<br>853       |  |
| 854              |  |
| 855<br>856       |  |
| 857              |  |
| 858<br>859       | request the Association to deliver to Ruyer before Clasing Date (63) a current Status Letter   |
| 860              | 16.4. Other Prorations. Water and sewer charges, propane, interest on continuing loan, and NONE.   |
| 861              | <b>16.5.</b> Final Settlement. Unless otherwise agreed in writing, these prorations shall be final.  |
| 862<br>863       | 47 P000F00ION P  |
| 864              | 17. POSSESSION. Possession of the Property shall be delivered to Buyer on Possession Date (§ 3) at   |
| 865<br>866       | Possession Time (§ 3), subject to the following leases or tenancies:  NONE   |
| 867              |  |
| 868<br>869       | If Seller, after Closing, fails to deliver possession as specified, Seller shall be subject to eviction and shall be additionally liable to Buyer for payment of \$ N/A per day (or any part of a day notwithstanding § 18.1.) from  |
| 870              | Possession Date(§ 3) and Possession Time (§ 3) until possession is delivered.  |
| 871<br>872       | r decession bate(3.0) and r desession time (3.0) and possession is delivered.  |
| 873<br>874       | GENERAL PROVISIONS   |
| 875              | GENERAL PROPERTY.  |
| 876<br>877       | 18. DAY; COMPUTATION OF PERIOD OF DAYS, DEADLINE.  |
| 878              | 18.1. Day. As used in this Contract, the term "day" shall mean the entire day ending at 11:59 p.m., United   |
| 879              | States Mountain Time (Standard or Daylight Savings as applicable).   |
| 880<br>881       | 18.2. Computation of Period of Days, Deadline. In computing a period of days, when the ending date   |
| 882              | is not specified, the first day is excluded and the last day is included, e.g., three days after MEC. If any deadline  |
| 883<br>884       | falls on a Saturday, Sunday or federal or Colorado state holiday (Holiday), such deadline Shall Shall Not  |
| 885              | be extended to the next day that is not a Saturday, Sunday or Holiday. Should neither box be checked, the  |
| 886<br>887       | deadline shall not be extended.  |
| 888              | 19. CAUSES OF LOSS, INSURANCE; DAMAGE TO INCLUSIONS AND SERVICES; CONDEMNATION;  |
| 889<br>890       | AND WALK—THROUGH. Except as otherwise provided in this Contract, the Property, Inclusions or both shall be   |
| 891              | delivered in the condition existing as of the date of this Contract, ordinary wear and tear excepted.  |
| 892<br>893       | 19.1. Causes of Loss, Insurance. In the event the Property or Inclusions are damaged by fire, other  |
| 894              | perils or causes of loss prior to Closing in an amount of not more than ten percent of the total Purchase Price  |
| 895<br>896       | (Property Damage), Seller shall be obligated to repair the same before Closing Date (§ 3). Buyer has the Right to  |
| 897              | Terminate under § 25.1, on or before <b>Closing Date</b> (§ 3), if the Property Damage is not repaired before <b>Closing</b>   |
| 898<br>899       | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such   |
| 900              | Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by   |
| 901<br>902       | Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the  |
| 903              | event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the  |
| 904<br>905       | Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the   |
| 905              | amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  |
| 907              | 19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and  |
| 908              | communication services), system, component or fixture of the Property (collectively Service), e.g., heating or   |
| 910              | plumbing, fail or be damaged between the date of this Contract and Closing or possession, whichever shall be   |
| 911<br>912       | earlier, then Seller shall be liable for the repair or replacement of such Inclusion or Service with a unit of similar   |
| 913              | size, age and quality, or an equivalent credit, but only to the extent that the maintenance or replacement of such   |
| 914<br>CDC4 0 12 | Inclusion or Service is not the responsibility of the Association, if any, less any insurance proceeds received by   |
| CBS4-9-12.       | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 11 of 15  |
|                  |  |
| Buyer(s) Init    | Sals Sallar(a) Initials  |
| Talon (a) mill   | als Seller(s) Initials   |

CTMeContracts.com - ©2013 CTM Software Corp.

937

939

942

943

945

946 947

948

949 950 951

952

953 954

955

956

957

958

959

960

961 962

963

964

966

967

968

969

973

974

975

976

977 978 979

980

981 982

983 984

985 986

987

988

989 990

992

993 994

995 996

997 998

Buyer(s) Initials\_ Seller(s) Initials

CONTRACT TO BUY AND SELL REAL ESTATE - Land

EARNEST MONEY DISPUTE. Except as otherwise provided herein, Earnest Money Holder shall release

Page 12 of 15

the Earnest Money as directed by written mutual instructions, signed by both Buyer and Seller. In the event of any

controversy regarding the Earnest Money (notwithstanding any termination of this Contract), Earnest Money Holder shall not be required to take any action. Earnest Money Holder, at its sole subjective discretion, has several options: (1) await any proceeding, (2) interplead all parties and deposit Earnest Money into a court of competent jurisdiction and shall recover court costs and reasonable attorney and legal fees, or (3) provide notice to Buyer and Seller that unless Earnest Money Holder receives a copy of the Summons and Complaint or Claim (between Buyer and Seller) containing the case number of the lawsuit (Lawsuit) within one hundred twenty days of Earnest Money Holder's notice to the parties, Earnest Money Holder shall be authorized to return the Earnest Money to Buyer. In the event Earnest Money Holder does receive a copy of the Lawsuit, and has not interplead the monies at the time of any Order, Earnest Money Holder shall disburse the Earnest Money pursuant to the Order of the Court. The parties reaffirm the obligation of **Mediation** (§ 23).

#### 25. TERMINATION.

- **25.1. Right to Terminate.** If a party has a right to terminate, as provided in this Contract (Right to Terminate), the termination shall be effective upon the other party's receipt of a written notice to terminate (Notice to Terminate), provided such written notice was received on or before the applicable deadline specified in this Contract. If the Notice to Terminate is not received on or before the specified deadline, the party with the Right to Terminate shall have accepted the specified matter, document or condition as satisfactory and waived the Right to Terminate under such provision.
- **25.2. Effect of Termination.** In the event this Contract is terminated, all Earnest Money received hereunder shall be returned and the parties shall be relieved of all obligations hereunder, subject to §§ 10.4, 22, 23 and 24.
- 26. ENTIRE AGREEMENT, MODIFICATION, SURVIVAL. This Contract, its exhibits and specified addenda, constitute the entire agreement between the parties relating to the subject hereof, and any prior agreements pertaining thereto, whether oral or written, have been merged and integrated into this Contract. No subsequent modification of any of the terms of this Contract shall be valid, binding upon the parties, or enforceable unless made in writing and signed by the parties. Any right or obligation in this Contract that, by its terms, exists or is intended to be performed after termination or Closing shall survive the same.

#### 27. NOTICE, DELIVERY, AND CHOICE OF LAW.

- 27.1. Physical Delivery. All notices must be in writing, except as provided in § 27.2. Any document, including a signed document or notice, from or on behalf of Seller, and delivered to Buyer shall be effective when physically received by Buyer, any signatory on behalf of Buyer, any named individual of Buyer, any representative of Buyer, or Brokerage Firm of Broker working with Buyer (except for delivery, after Closing, of the notice requesting mediation described in § 23) and except as provided in § 27.2. Any document, including a signed document or notice, from or on behalf of Buyer, and delivered to Seller shall be effective when physically received by Seller, any signatory on behalf of Seller, any named individual of Seller, any representative of Seller, or Brokerage Firm of Broker working with Seller (except for delivery, after Closing, of the notice requesting mediation described in § 23) and except as provided in § 27.2.
- **27.2.** Electronic Delivery. As an alternative to physical delivery, any document, including any signed document or written notice, may be delivered in electronic form only by the following indicated methods: Facsimile E-mail Internet In No Electronic Delivery. If the box "No Electronic Delivery" is checked, this § 27.2 shall not be applicable and § 27.1 shall govern notice and delivery. Documents with original signatures shall be provided upon request of any party.
- **27.3.** Choice of Law. This Contract and all disputes arising hereunder shall be governed by and construed in accordance with the laws of the State of Colorado that would be applicable to Colorado residents who sign a contract in Colorado for property located in Colorado.
- **28. NOTICE OF ACCEPTANCE, COUNTERPARTS.** This proposal shall expire unless accepted in writing, by Buyer and Seller, as evidenced by their signatures below, and the offering party receives notice of such acceptance pursuant to § 27 on or before **Acceptance Deadline Date** (§ 3) and **Acceptance Deadline Time** (§ 3). If accepted, this document shall become a contract between Seller and Buyer. A copy of this Contract may be executed by each party, separately, and when each party has executed a copy thereof, such copies taken together shall be deemed to be a full and complete contract between the parties.
- 29. GOOD FAITH. Buyer and Seller acknowledge that each party has an obligation to act in good faith, including but not limited to exercising the rights and obligations set forth in the provisions of Financing Conditions and Obligations (§ 5), Record Title and Off-Record Title (§ 8), Current Survey Review (§ 9) and Property Disclosure, Inspection, Indemnity, Insurability, Due Diligence, Buyer Disclosure and Source of Water (§ 10).

| Disclosure, Inspection, Indemnity, Insurability, Due Diligence, Buyer Disclosure and Source of Water 1078 10).  ADDITIONAL PROVISIONS AND ATTACHMENTS | er ( |
|---|------|
| CBS4-9-12. CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 13 of 15  |      |
| Buyer(s) Initials Seller(s) Initials  |      |
| CTMeContracts.com - ©2013 CTM Software Corp.  |      |

| 1084   | ,   |
|--|---|
| 1085<br>1086<br>1087<br>1088                         | 1. SELLER REPRESENTS THE PROPERTY IS NOT SUBJECT TO ANY MORATORIUM THAT WOULD   |
| 1089<br>1090<br>1091<br>1092<br>1093<br>1094         | 2. SELLER REPRESENTS THAT TO ITS KNOWLEDGE THERE ARE NO ADVERSE ENVIRONMENTAL CONDITIONS THAT WOULD PREVENT BUYER FROM UTILIZING THE PROPERTY FOR ITS INTENDED USE FOR THE CONSTRUCTION OF A SINGLE FAMILY RESIDENCE.   |
| 1095<br>1096<br>1097<br>1098<br>1099<br>1100         | 3. BUYER REPRESENTS HE IS A COLORADO LICENSED REAL ESTATE AGENT AND MANAGER OF IRON KEY HOMES WITH THE FULL AUTHORITY TO EXECUTE ANY AND ALL DOCUMENTS TO EFFECTUATE THE CONSUMMATION OF THIS TRANSACTION.  |
| 1101<br>1102<br>1103                                 | 4. IT IS FURTHER AGREED BY THE PARTIES THAT MEC, MUTUAL EXECUTION OF CONTRACT, BEGINS FOLLOWING CITY COUNCIL APPROVAL OF THIS AGREEMENT.  |
| 1109<br>1110<br>1111<br>1112                         | Declaration of Covenants, Conditions and Restrictions of Canyon View, as are more particularly found at Reception Nos. 2007006559, and 2007030761, of the records of the Jefferson County   |
| 1113<br>1114<br>1115                                 | 31.2. The following disclosure forms are attached but are not a part of this Contract: n/a  |
| 1116<br>1117   |   |
| 1118<br>1119   | SIGNATURES  |
| 1120<br>1121<br>1122                                 | Buyer: Iron Key Homes, LLC  |
| 1123<br>1124   | By: Douglas Ferraro, Managing Partner   |
| 1125<br>1126   | Address:16549 W 15TH AVE Golden CO 80401 Phone: 3038887436 Fax: 3032735694  |
| 1127<br>1128   | Electronic Address: doug@ironkeyhomes.com   |
| 1129<br>1130<br>1131<br>1132<br>1133<br>1134<br>1135 | [NOTE: If this offer is being countered or rejected, do not sign this document. Refer to $\S$ 32]   |
| 1136<br>1137   | Date:<br>Seller: <i>City of Golden</i>  |
| 1138<br>1139   | By: City of Golden  |
| 1140<br>1141   | Address: 911 10th Street Golden CO 80401 Phone: 303.384.8097 Fax:   |
| 1142<br>1143<br>1144                                 | Electronic Address:   |
| 1144<br>1145<br>1146                                 |   |
| 1147<br>1148<br>1149                                 | 32. COUNTER; REJECTION. This offer is ☐ Countered ☐ Rejected. Initials only of party (Buyer or Seller) who countered or rejected offer  |
| 1150<br>1151   | City of Golden  |
| 1152<br>1153   | By: City of Golden  |
| 1154<br>1155   | END OF CONTRACT TO BUY AND SELL REAL ESTATE   |
| 1156<br>1157   | 22 PROVERS AGVANOMI FROMENTO AND COMPENSATION RICCI COURT   |
| 1158<br>1159<br>1160                                 | 33. BROKER'S ACKNOWLEDGMENTS AND COMPENSATION DISCLOSURE. (To be completed by Broker working with Buyer)  |
| 1164<br>1165<br>1166                                 | Broker Does Does Not acknowledge receipt of Earnest Money deposit and, while not a party to the Contract, agrees to cooperate upon request with any mediation concluded under § 23. Broker agrees that if Brokerage Firm is the Earnest Money Holder and, except as provided in § 24, if the Earnest Money has not already been returned following receipt of a Notice to Terminate or other written notice of termination, Earnest |
| CBS4-9-12. C   | ONTRACT TO BUY AND SELL REAL ESTATE - Land Page 14 of 15  |
| Buyer(s) Initial                                     | s Seller(s) Initials  |

| 116<br>116<br>116<br>117<br>117 | Earnest Money shall be made within five days of Earnest Money Holder's receipt of the executed written mutual instructions, provided the Earnest Money check has cleared.   |  |  |  |  |  |  |
|---------------------------------|---|--|--|--|--|--|--|
| 117<br>117                      |   |  |  |  |  |  |  |
|                                 | Brokerage Firm's compensation or commission is to be paid by ☐ <b>Listing Brokerage Firm</b> ☐ <b>Buyer</b> ☐ <b>Other</b> <i>NONE - NO COMMISSION DUE</i>  |  |  |  |  |  |  |
|                                 | Brokerage Firm's Name: Colorado Land Professionals, LLC Date: 2/19/3 Broker's Name: Douglas Ferraro Address: 1505 Quaker Street Golden, CO 80401 Ph: 303.888.7436 Fax: 303.273.5694 Email: coloradolandpros@gmail.com   |  |  |  |  |  |  |
|                                 | 34. BROKER'S ACKNOWLEDGMENTS AND COMPENSATION DISCLOSURE. (To be completed by Broker working with Seller)   |  |  |  |  |  |  |
|                                 | Broker Does Does Not acknowledge receipt of Earnest Money deposit specified in § 4.1 and, while not a party to the Contract, agrees to cooperate upon request with any mediation concluded under § 23. Broker agrees that if Brokerage Firm is the Earnest Money Holder and, except as provided in § 24, if the Earnest Money has not already been returned following receipt of a Notice to Terminate or other written notice of termination, Earnest Money Holder shall release the Earnest Money as directed by the written mutual instructions. Such release of Earnest Money shall be made within five days of Earnest Money Holder's receipt of the executed written mutual instructions, provided the Earnest Money check has cleared. |  |  |  |  |  |  |
|                                 | Broker is working with the Seller as a $\square$ Seller's Agent $\square$ Buyer's Agent $\square$ Transaction-Broker in this transaction. $\square$ This is a Change of Status.   |  |  |  |  |  |  |
|                                 | Brokerage Firm's compensation or commission is to be paid by Seller Buyer Other n/a  Brokerage Firm's Name: n/a  Broker Date:   |  |  |  |  |  |  |
|                                 | Address: <i>n/a n/a, n/a n/a</i> Ph: n/a Fax: n/a Email:  |  |  |  |  |  |  |
|                                 | CBS4-9-12. CONTRACT TO BUY AND SELL REAL ESTATE (LAND) CTM eContracts - ©2012 CTM Software Corp.  |  |  |  |  |  |  |
|                                 |   |  |  |  |  |  |  |
|                                 |   |  |  |  |  |  |  |
|                                 |   |  |  |  |  |  |  |
|                                 |   |  |  |  |  |  |  |
|                                 |   |  |  |  |  |  |  |
| CBS4-9-12.                      | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 15 of 15   |  |  |  |  |  |  |
| Buyer(s) Initi                  | als Seller(s) Initials  |  |  |  |  |  |  |

# Res. 2249

# Colorado Land Professionals, LLC

Douglas Ferraro

Ph: 303.888.7436 Fax: 303.273.5694

The printed portions of this form, except differentiated additions, have been approved by the Colorado Real Estate Commission. (CBS4-9-12) (Mandatory 1-13)

THIS FORM HAS IMPORTANT LEGAL CONSEQUENCES AND THE PARTIES SHOULD CONSULT LEGAL AND TAX OR OTHER COUNSEL BEFORE SIGNING.

# CONTRACT TO BUY AND SELL REAL ESTATE (LAND)

( Property with No Residences)

|  |   | ( Property with Residences-Residential Add   |  |
|--|---|--|--|
|  |   |  | Date: 2/19/201   |
|  |   | AGREEMENT  |  |
| 1.<br>Property                         | AGREEM<br>y described                                       | ENT. Buyer, identified in § 2.1, agrees to buy, and Sel below on the terms and conditions set forth in this cor  | ller, identified in § 2.3, agrees to sell ntract (Contract).   |
| 2.                                     | PARTIES /   | AND PROPERTY.  |  |
|  |   | r. Buyer, Iron Key Homes, LLC, will take title to the  | Bronothy deposits at the target of   |
| Tenants                                | Tena  | nts In Common A Other Limited Liability Comp   | nanv   |
|  | 2.2. Assig  | gnability and Inurement. This Contract Shall   | Shall Not be assignable by Ruger   |
| without \$                             | Seller's prio   | r written consent. Except as so restricted, this Contract  | ct shall inure to the benefit of and be  |
| binding (                              | upon the ne   | irs, personal representatives, successors and assigns  | of the parties   |
|  | 2.3. Seller   | Seller, City of Golden, is the current owner of the  | Property described below   |
| latta                                  | 2.4. Prope  | <b>rry.</b> The Property is the following legally described rea  | al estate in the County of   |
| Jeffers                                | on  | . Colorado:  | •  |
| known -                                | DLUCK 2,  | FILING NO. 2, CANYON VIEW SUBDIVISION  |  |
| known a                                | S NO. 826 .   | loesph Circle Golden CO 80401  |  |
| and all in                             | with the int  | erests, easements, rights, benefits, improvements and  | dattached fixtures appurtenant there   |
| and an n                               | 25 Inclue   | eller in vacated streets and alleys adjacent thereto, exc  | cept as herein excluded (Property).  |
|  | 251   | ions. The Purchase Price includes the following items  | s (Inclusions):  |
| Other Fi                               | xtures: NC  | Fixtures. All fixtures attached to the Property on the   | date of this Contract.   |
|  |   | rached to the Property after the date of this Contract, s  | nuch additional financia   |
| in the Pu                              | rchase Pric   | e.   | such additional fixtures are also inclu  |
|  | 2.5.2.  |  | ched or not on the date of this Contr  |
| the follow                             | ving items a  | re included:   | and a net on the date of the bond  |
| NONE                                   |   |  |  |
| Other Pe                               | ersonal Pro   | perty:   |  |
| NONE                                   |   |  |  |
|  | The Pers  |  |  |
|  | cept persor   | ional Property to be conveyed at Closing shall be con-   | veyed by Seller free and clear of all  |
| axes (ex                               |   | sonal Property to be conveyed at Closing shall be con-<br>nal property taxes for the year of Closing), liens and er  | ncumbrances, except  |
| taxes (ex<br><i>NONE</i>               | · · · · · · · · · · · · · · · · · · ·                       | lal property taxes for the year of Closing), liens and er  | ncumbrances, except<br>le or other applicable legal instrumer  |
| NUNE                                   | 2.5.3.  | lal property taxes for the year of Closing), liens and er  | ncumbrances, except<br>le or other applicable legal instrumer  |
| taxes (ex<br>NONE<br>NONE              | 2.5.3.  | nai property taxes for the year of Closing), liens and er  Conveyance shall be by bill of sal  Trade Fixtures. With respect to trade fixtures, Selle   | ncumbrances, except<br>le or other applicable legal instrumen<br>er and Buyer agree as follows:  |
| NONE                                   | 2.5.3.  | Conveyance shall be by bill of sales and erms. With respect to trade fixtures, Selles Fixtures to be conveyed at Closing shall be conveyed.  | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows:  |
| NONE  except p                         | 2.5.3.  | lal property taxes for the year of Closing), liens and er  | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows:  |
| NONE<br>NONE<br>(except po             | 2.5.3. The Tradersonal projects                             | La property taxes for the year of Closing), liens and er Conveyance shall be by bill of sale Trade Fixtures. With respect to trade fixtures, Selle le Fixtures to be conveyed at Closing shall be conveyed perty taxes for the year of Closing), liens and encumber  | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows:  |
| NONE<br>PEXCEPT PONE                   | 2.5.3.  The Tradersonal proposes shall be                   | Le Fixtures to be conveyed at Closing), liens and er conveyed at Closing shall be conveyed er fixtures. With respect to trade fixtures, Selle le Fixtures to be conveyed at Closing shall be conveyed erry taxes for the year of Closing), liens and encumbing by bill of sale or other applicable legal instrument.   | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows:  |
| NONE<br>NONE<br>(except po             | 2.5.3. The Tradersonal projects                             | La property taxes for the year of Closing), liens and er Conveyance shall be by bill of sale Trade Fixtures. With respect to trade fixtures, Selle le Fixtures to be conveyed at Closing shall be conveyed perty taxes for the year of Closing), liens and encumber  | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows:  |
| NONE<br>(except po<br>NONE<br>Conveyar | 2.5.3.  The Tradersonal proposes shall be                   | Le Fixtures to be conveyed at Closing shall be conveyed by bill of sale or other applicable legal instrument.  Water Rights, Water and Sewer Taps.   | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows: ed by Seller free and clear of all taxe rances, except                             |
| NONE<br>NONE<br>(except po             | 2.5.3.  The Tradersonal proposes shall be                   | Le Fixtures to be conveyed at Closing), liens and er conveyed at Closing shall be conveyed er fixtures. With respect to trade fixtures, Selle le Fixtures to be conveyed at Closing shall be conveyed erry taxes for the year of Closing), liens and encumbing by bill of sale or other applicable legal instrument.   | ncumbrances, except le or other applicable legal instrument er and Buyer agree as follows: ed by Seller free and clear of all taxe rances, except                            |
| NONE (except points) NONE Conveyar     | 2.5.3.  The Tradersonal projects shall be 2.5.4.  Any water | Le Fixtures to be conveyed at Closing), liens and er conveyed at Closing shall be conveyed error taxes for the year of Closing shall be conveyed error taxes for the year of Closing), liens and encumber by bill of sale or other applicable legal instrument.  Water Rights, Water and Sewer Taps.  2.5.4.1. Deeded Water Rights. The following legaling the conveyed at Closing shall be conveyed at C | ncumbrances, except le or other applicable legal instrument er and Buyer agree as follows: ed by Seller free and clear of all taxe rances, except by described water rights: |
| NONE<br>except po<br>NONE<br>Conveyar  | 2.5.3.  The Tradersonal projects shall be 2.5.4.  Any water | Le Fixtures to be conveyed at Closing shall be conveyed by bill of sale or other applicable legal instrument.  Water Rights, Water and Sewer Taps.   | ncumbrances, except le or other applicable legal instrumer er and Buyer agree as follows: ed by Seller free and clear of all taxe rances, except                             |

CONTRACT TO BUY AND SELL REAL ESTATE - Land

Page 1 of 15

Buyer(s) Initials

10

12

14 15

78 80 81

Seller(s) Initials

| to or at Clos<br>the Colorado<br>a registration<br>closing serv | sing, con<br>Division<br>of existice in co | rmation about such well to Buyer. Buyer understands that if the well to be transferred is a or a Domestic Exempt Water Well used for ordinary household purposes, Buyer shall, prior a Change in Ownership form for the well. If an existing well has not been registered with not Water Resources in the Department of Natural Resources (Division), Buyer shall complete ting well form for the well and pay the cost of registration. If no person will be providing a nnection with the transaction, Buyer shall file the form with the Division within sixty days after ermit # is WA |
|---|--|--|
| NONE  |  | 2.5.4.3.  Water Stock Certificates:  |
|   |  | 2.5.4.4. ☐ Water Tap ☐ Sewer Tap   |
| Note: Buye<br>paid, if any,                                     | r is adv<br>time a                         | sed to obtain, from the provider, written confirmation of the amount remaining to be not other restrictions for transfer and use of the tap.   |
| NONE  |  | 2.5.4.5. Other Rights:   |
| NONE  | 2.5.5.                                     | Growing Crops. With respect to growing crops, Seller and Buyer agree as follows:   |

**2.6. Exclusions.** The following items are excluded (Exclusions):

## NONE

# 3. DATES AND DEADLINES.

| Item No.    | Reference | Event  | Date or Deadline |
|-------------|-----------|--|------------------|
| 1           | § 4.3     | Alternative Earnest Money Deadline                 | MARCH 5TH, 2013  |
|             |           | Title and Association                              |                  |
| 2           | § 7.1     | Record Title Deadline                              | MARCH 6TH, 2013  |
| 3           | § 7.5     | Exceptions Request Deadline                        | MARCH 6TH, 2013  |
| 4           | § 8.1     | Record Title Objection Deadline                    | MARCH 21ST, 2013 |
| 5           | § 8.2     | Off-Record Title Deadline                          | MARCH 6TH, 2013  |
| <u>6</u> 7  | § 8.2     | Off-Record Title Objection Deadline                | MARCH 21ST, 2013 |
|             | § 8.3     | Title Resolution Deadline                          | MARCH 26TH, 2013 |
| 8           | § 7.6     | Association Documents Deadline                     | MARCH 6TH, 2013  |
| 9           | § 7.6     | Association Documents Objection Deadline           | MARCH 21ST, 2013 |
| 10          | § 8.5     | Right of First Refusal Deadline                    | NA               |
|             |           | Seller's Property Disclosure                       |                  |
| 11          | § 10.1    | Seller's Property Disclosure Deadline              | NA               |
|             |           | Loan and Credit                                    |                  |
|             | § 5.1     | Loan Application Deadline                          | NA               |
| 13          | § 5.2     | Loan Objection Deadline                            | N/A              |
| 14          | § 5.3     | Buyer's Credit Information Deadline                | N/A              |
|             | § 5.3     | Disapproval of Buyer's Credit Information Deadline | N/A              |
|             | § 5.4     | Existing Loan Documents Deadline                   | N/A              |
|             | § 5.4     | Existing Loan Documents Objection Deadline         | N/A              |
| 18          | § 5.4     | Loan Transfer Approval Deadline                    | NA               |
| <del></del> |           | Appraisal  |                  |
|             | § 6.2     | Appraisal Deadline                                 | N/A              |
| 20          | § 6.2     | Appraisal Objection Deadline                       | N/A              |
|             |           | Survey   |                  |
| 21          | § 9.1     | Current Survey Deadline                            | N/A              |
| 22          | § 9.2     | Current Survey Objection Deadline                  | NA               |
|             |           | nspection and Due Diligence                        |                  |
| 23          | § 10.2    | Inspection Objection Deadline                      | APRIL 7th, 2013  |
| 24          | § 10.3    | Inspection Resolution Deadline                     | APRIL 11TH, 2013 |
| 25          | § 10.5    | Property Insurance Objection Deadline              | NA               |
| 26          | § 10.6    | Due Diligence Documents Delivery Deadline          | MARCH 5TH, 2013  |

CBS4-9-12. CONTRACT TO BUY AND SELL REAL ESTATE - Land

Page 2 of 15

| Buyer(s) | Initials |
|----------|----------|
|----------|----------|

Seller(s) Initials

| 27 | § 10.7 | Due Diligence Documents Objection Deadline    | APRIL 1ST, 2013               |
|----|--------|---|-------------------------------|
| 28 | § 10.8 | Environmental Inspection Objection Deadline   | APRIL 11TH, 2013              |
| 29 | § 10.8 | ADA Evaluation Objection Deadline             | N/A                           |
| 30 | § 11.1 | Tenant Estoppel Statements Deadline           | N/A                           |
| 31 | § 11.2 | Tenant Estoppel Statements Objection Deadline | N/A                           |
|    |        | Closing and Possession                        |                               |
| 32 | § 12.3 | Closing Date                                  | APRIL 12TH, 2013              |
| 33 | § 17   | Possession Date                               | DOD                           |
| 34 | § 17   | Possession Time                               | IMMEDIATLEY FOLLOWING CLOSING |
| 35 | § 28   | Acceptance Deadline Date                      | MARCH 1ST, 2013               |
| 36 | § 28   | Acceptance Deadline Time                      | 5:00 P.M                      |
| 37 | n/a    | n/a   | n/a                           |
| 38 | n/a    | n/a   | n/a                           |

Note: Applicability of Terms.

Any box, blank or line in this Contract left blank or completed with the abbreviation "N/A", or the word "Deleted" means such provision in **Dates and Deadlines** (§ 3), including any deadline, is not applicable and the corresponding provision of this Contract to which reference is made is deleted.

The abbreviation "MEC" (mutual execution of this Contract) means the date upon which both parties have signed this Contract.

#### PURCHASE PRICE AND TERMS.

**4.1. Price and Terms.** The Purchase Price set forth below shall be payable in U.S. Dollars by Buyer as follows:

| Item No. | Reference | Item                        | Amount       | Amount       |
|----------|-----------|-----------------------------|--------------|--------------|
| 1        | § 4.1     | Purchase Price              | \$100,000.00 |              |
| 2        | § 4.2     | Earnest Money               |              | \$5,000.00   |
| 3        | § 4.5     | New Loan                    |              |              |
| 4        | § 4.6     | Assumption Balance          |              |              |
| 5        | § 4.7     | Seller or Private Financing |              |              |
| 6        | n/a       | n/a                         |              |              |
| 7        | n/a       | n/a                         |              |              |
| 8        | § 4.3     | Cash at Closing             |              | \$95,000.00  |
| 9        |           | TOTAL                       | \$100,000.00 | \$100,000.00 |

**Seller Concession.** Seller, at Closing, shall credit, as directed by Buyer, an amount of to assist with any and all of the following: Buyer's closing costs, (Seller Concession). Seller Concession is in addition to any sum Seller has agreed to pay or credit Buyer elsewhere in this Contract. Seller Concession will be reduced to the extent it exceeds the aggregate of what is allowed by Buyer's lender as set forth in the Closing Statement, Closing Disclosure or HUD-1, at Closing.

**4.3. Earnest Money.** The Earnest Money set forth in this section, in the form of **COMPANY CHECK**, shall be payable to and held by **TITLE COMPANY** 

(Earnest Money Holder), in its trust account, on behalf of both Seller and Buyer. The Earnest Money deposit shall be tendered, by Buyer, with this Contract unless the parties mutually agree to an **Alternative Earnest Money Deadline** (§ 3) for its payment. The parties authorize delivery of the Earnest Money deposit to the company conducting the Closing (Closing Company), if any, at or before Closing. In the event Earnest Money Holder has agreed to have interest on Earnest Money deposits transferred to a fund established for the purpose of providing affordable housing to Colorado residents, Seller and Buyer acknowledge and agree that any interest accruing on the Earnest Money deposited with the Earnest Money Holder in this transaction shall be transferred to such fund.

**4.3.1.** Alternative Earnest Money Deadline. The deadline for delivering the Earnest Money, if other than at the time of tender of this Contract is as set forth as the Alternative Earnest Money Deadline (§ 3).

4.3.2. Return of Earnest Money. If Buyer has a Right to Terminate and timely terminates, Buyer shall be entitled to the return of Earnest Money as provided in this Contract. If this Contract is terminated as set forth in § 25 and, except as provided in § 24, if the Earnest Money has not already been returned following receipt of a Notice to Terminate, Seller agrees to execute and return to Buyer or Broker working with Buyer, written mutual instructions, i.e., Earnest Money Release form, within three days of Seller's receipt of such form.

Page 3 of 15

4.4. Form of Funds; Time of Payment; Funds Available.

CONTRACT TO BUY AND SELL REAL ESTATE - Land

| 1 -               | and C              |
|-------------------|--------------------|
|                   | ***1/H~V} \        |
| Buyer(s) Initials | Seller(s) Initials |
|                   |                    |

| 249               | 4.4.1. Good Funds. All amounts payable by the parties at Closing, including any loan   |
|-------------------|--|
| 250               | proceeds, Cash at Closing and closing costs, shall be in funds that comply with all applicable Colorado laws,                    |
| 251<br>252        | including electronic transfer funds, certified check, savings and loan teller's check and cashier's check (Good                  |
| 253               | Funds).  |
| 254               | 4.4.2. Available Funds. All funds required to be paid at Closing or as otherwise agreed in                                       |
| 255               | writing between the parties shall be timely paid to allow disbursement by Closing Company at Closing OR SUCH                     |
| 256<br>257        | PARTY CHAIL BE IN DESAULT During represents the Rouge of the date of this Contract Research                                      |
| 258               | PARTY SHALL BE IN DEFAULT. Buyer represents that Buyer, as of the date of this Contract, Does Does                               |
| 259               | Not have funds that are immediately verifiable and available in an amount not less than the amount stated as                     |
| 260               | Cash at Closing in § 4.1.  |
| 261<br>262<br>263 | 4.5. New Loan. (Omitted as inapplicable)   |
| 264<br>265        | 4.6. Assumption. (Omitted as inapplicable)   |
| 266<br>267        | 4.7. Seller or Private Financing. (Omitted as inapplicable)  |
| 268<br>269        | 4.7. Seliei di Fittate Filiancing. (Officied as mappilicable)  |
| 270               | TRANSACTION PROVISIONS   |
| 271               |  |
| 272               |  |
| 273<br>274        | 5. FINANCING CONDITIONS AND OBLIGATIONS. (Omitted as inapplicable)   |
| 275               | in interest delibriorie Arib Obligationic. (Crimical de mappinado)   |
| 276               | 5.0 Condit Interpretion and Revenue New Control Land (Continue on Inspection)  |
| 277<br>278        | 5.3. Credit Information and Buyer's New Senior Loan. (Omitted as inapplicable)   |
| 279<br>280        | 5.4. Existing Loan Review. (Omitted as inapplicable)   |
| 281               |  |
| 282               | 6. APPRAISAL PROVISIONS.   |
| 283<br>284        | 6.1. Lender Property Requirements. If the lender imposes any requirements or repairs (Requirements)                              |
| 285               | to be made to the Property (e.g., roof repair, repainting), beyond those matters already agreed to by Seller in this             |
| 286               |  |
| 287               | Contract, Seller has the Right to Terminate under § 25.1, (notwithstanding § 10 of this Contract), on or before three            |
| 288               | days following Seller's receipt of the Requirements, based on any unsatisfactory Requirements, in Seller's sole                  |
| 289<br>290        | subjective discretion. Seller's Right to Terminate in this § 6.1 shall not apply if, on or before any termination by             |
| 291               | Seller pursuant to this § 6.1: (1) the parties enter into a written agreement regarding the Requirements; or (2) the             |
| 292               | Requirements have been completed; or (3) the satisfaction of the Requirements is waived in writing by Buyer.                     |
| 293               | 6.2. Appraisal Condition. The applicable Appraisal provision set forth below shall apply to the                                  |
| 294<br>295        | respective loan type set forth in § 4.5.3, or if a cash transaction, i.e. no financing, § 6.2.1 shall apply.                     |
| 296               | <b>6.2.1.</b> Conventional/Other. Buyer has the sole option and election to terminate this Contract                              |
| 297               | if the Property's valuation, determined by an appraiser engaged on behalf of <b>N/A</b> is less than                             |
| 298<br>299        | the Purchase Price. The appraisal shall be received by Buyer or Buyer's lender on or before Appraisal Deadline                   |
| 300               | (§ 3). Buyer has the Right to Terminate under § 25.1, on or before Appraisal Objection Deadline (§ 3), if the                    |
| 301               | Property's valuation is less than the Purchase Price and Seller's receipt of either a copy of such appraisal or                  |
| 302               | written notice from lender that confirms the Property's valuation is less than the Purchase Price. This § 6.2.1 is for           |
| 303<br>304        | the sole benefit of Buyer.   |
| 305               | 6.3. Cost of Appraisal. Cost of any appraisal to be obtained after the date of this Contract shall be                            |
| 306               | timely paid by Buyer Seller. The cost of the appraisal may include any and all fees paid to the appraiser,                       |
| 307               |  |
| 308<br>309        | appraisal management company, lender's agent or all three.   |
| 310               |  |
| 311               | 7. EVIDENCE OF TITLE AND ASSOCIATION DOCUMENTS.  |
| 312               | 7.1.   |
| 313<br>314        | insurance company to furnish the owner's title insurance policy at Seller's expense. On or before Record Title                   |
| 315               | Deadline (§ 3), Seller shall furnish to Buyer a current commitment for owner's title insurance policy (Title                     |
| 316               | Commitment), in an amount equal to the Purchase Price, or if this box is checked $\square$ an <b>Abstract</b> of title certified |
| 317               | to a current date. Seller shall cause the title insurance policy to be issued and delivered to Buyer as soon as                  |
| 318<br>319        | practicable at or after Closing.   |
| 320               | 7.2. Buyer Selects Title Insurance Company. If this box is checked, Buyer shall select the                                       |
| 321               | title insurance company to furnish the owner's title insurance policy at Buyer's expense. On or before <b>Record</b>             |
| 322<br>323        |  |
| 323<br>324        | Title Deadline (§ 3), Buyer shall furnish to Seller, a current commitment for owner's title insurance policy (Title              |
| 325               | Commitment), in an amount equal to the Purchase Price.   |
| 326               | If neither box in § 7.1 or § 7.2 is checked, § 7.1 applies.  |
| 327<br>328        | 7.3. Owner's Extended Coverage (OEC). The Title Commitment Shall Shall Shall Not commit  |
| 329               | to delete or insure over the standard exceptions which relate to: (1) parties in possession, (2) unrecorded                      |
| 330               | easements, (3) survey matters, (4) unrecorded mechanics' liens, (5) gap period (effective date of commitment to                  |
| 331               | date deed is recorded), and (6) unpaid taxes, assessments and unredeemed tax sales prior to the year of Closing                  |
| CBS4-9-12.        | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 4 of 15   |
|                   | m c  |
| Buyer(s) Init     | ialsSeller(s) Initials   |

 $\label{lem:com-com-contracts} \textbf{CTMeContracts.com-} \textbf{@2013 CTM Software Corp.}$ 

| 332        | (OEC).   |
|------------|--|
| 333<br>334 | Note: The title insurance company may not agree to delete or insure over any or all of the standard exceptions.            |
| 335        | 7.3.1. Premium for OEC.If the title insurance company agrees to provide an endorsement for                                 |
| 336        | OEC, any additional premium expense to obtain an endorsement for OEC shall be paid by Buyer   Seller                       |
| 337        |  |
| 338<br>339 | — one man by buyer and one man by coner — one.   |
| 340        | 7.4. Buyer's Right to Review Title Commitment and Title Documents. Buyer has the right to                                  |
| 341        | review the Title Commitment, its provisions and Title Documents (defined in § 7.5), and if not satisfactory to Buyer,      |
| 342        | Buyer may exercise Buyer's rights pursuant to § 8.1.   |
| 343        | 7.5. Copies of Exceptions. Unless the box in § 7.2 is checked (Buyer Selects Title Insurance                               |
| 344<br>345 | Company) on or before Record Title Deadline (§ 3), Seller, at Seller's expense, shall furnish to Buyer and                 |
| 346        | , (1) copies of any plats,   |
| 347        | declarations, covenants, conditions and restrictions burdening the Property, and (2) if a Title Commitment is              |
| 348        | required to be furnished, and if this box is checked $\square$ Copies of any Other Documents (or, if illegible,            |
| 349        |  |
| 350<br>351 | summaries of such documents) listed in the schedule of exceptions (Exceptions). Even if the box is not checked,            |
| 352        | Seller shall have the obligation to furnish these documents pursuant to this section if requested by Buyer any time        |
| 353        | on or before Exceptions Request Deadline (§ 3). This requirement shall pertain only to documents as shown of               |
| 354        | record in the office of the clerk and recorder in the county where the Property is located. The Abstract or Title          |
| 355        | Commitment, together with any copies or summaries of such documents furnished pursuant to this section,                    |
| 356        | constitute the title documents (collectively, Title Documents).  |
| 357<br>358 |  |
| 359        | 7.5.1 Existing Abstracts of Title. Seller shall deliver to Buyer copies of any abstracts of title                          |
| 360        | covering all or any portion of the Property (Abstract) in Seller's possession on or before <b>Record Title Deadline</b> (§ |
| 361        | 3).  |
| 362        | <b>7.6.</b> Homeowners' Association Documents. Homeowners' Association Documents (Association                              |
| 363<br>364 | Documents) consist of the following:   |
| 365        | 7.6.1. All Homeowners' Association declarations, bylaws, operating agreements, rules and                                   |
| 366        | regulations, party wall agreements.  |
| 367        | 7.6.2. Minutes of most recent annual owners' meeting.  |
| 368        | 7.6.3. Minutes of any directors' or managers' meetings during the six-month period immediately                             |
| 369<br>370 | preceding the date of this Contract. If none of the preceding minutes exist, then the most recent minutes, if any          |
| 371        |  |
| 372        | (§§ 7.6.1, 7.6.2 and 7.6.3, collectively, Governing Documents).  |
| 373        | 7.6.4. The most recent financial documents which consist of: (1) annual and most balance                                   |
| 374        | sheet, (2) annual and most recent income and expenditures statement, (3) annual budget, and (4) reserve study, if          |
| 375<br>376 | any (collectively, Financial Documents).   |
| 377        | 7.6.5. Common Interest Community Disclosure. THE PROPERTY IS LOCATED WITHIN A  |
| 378        | COMMON INTEREST COMMUNITY AND IS SUBJECT TO THE DECLARATION FOR SUCH COMMUNITY. THE  |
| 379        | OWNER OF THE PROPERTY WILL BE REQUIRED TO BE A MEMBER OF THE OWNER'S ASSOCIATION FOR                                       |
| 380        | THE COMMUNITY AND WILL BE SUBJECT TO THE BYLAWS AND RULES AND REGULATIONS OF THE   |
| 381<br>382 |  |
| 383        | ASSOCIATION. THE DECLARATION, BYLAWS, AND RULES AND REGULATIONS WILL IMPOSE FINANCIAL                                      |
| 384        | OBLIGATIONS UPON THE OWNER OF THE PROPERTY, INCLUDING AN OBLIGATION TO PAY   |
| 385        | ASSESSMENTS OF THE ASSOCIATION. IF THE OWNER DOES NOT PAY THESE ASSESSMENTS, THE   |
| 386        | ASSOCIATION COULD PLACE A LIEN ON THE PROPERTY AND POSSIBLY SELL IT TO PAY THE DEBT.                                       |
| 387<br>388 | THE DECLARATION, BYLAWS, AND RULES AND REGULATIONS OF THE COMMUNITY MAY PROHIBIT THE                                       |
| 389        | OWNER FROM MAKING CHANGES TO THE PROPERTY WITHOUT AN ARCHITECTURAL REVIEW BY THE   |
| 390        | ASSOCIATION (OR A COMMITTEE OF THE ASSOCIATION) AND THE APPROVAL OF THE ASSOCIATION.                                       |
| 391        | PURCHASERS OF PROPERTY WITHIN THE COMMON INTEREST COMMUNITY SHOULD INVESTIGATE THE   |
| 392        | FINANCIAL OBLIGATIONS OF MEMBERS OF THE ASSOCIATION, PURCHASERS SHOULD CAREFULLY   |
| 393<br>394 |  |
| 395        | READ THE DECLARATION FOR THE COMMUNITY AND THE BYLAWS AND RULES AND REGULATIONS OF   |
| 396        | THE ASSOCIATION.   |
| 397        | 7.6.6. Association Documents to Buyer.   |
| 398        | 7.6.6.1. Seller to Provide Association Documents. Seller shall cause the Association                                       |
| 399<br>400 | Documents to be provided to Buyer, at Seller's expense, on or before Association Documents Deadline (§ 3).                 |
| 401        | 7.6.6.2. Seller Authorizes Association. Seller authorizes the Association to provide the                                   |
| 402        | Association Documents to Buyer, at Seller's expense.   |
| 403        | 7.6.6.3. Seller's Obligation. Seller's obligation to provide the Association Documents                                     |
| 404        |  |
| 405<br>406 | shall be fulfilled upon Buyer's receipt of the Association Documents, regardless of who provides such documents.           |
| 407        | Note: If neither box in this § 7.6.6 is checked, the provisions of § 7.6.6.1 shall apply.                                  |
| 408        | 7.6.7. Conditional on Buyer's Review. Buyer has the right to review the Association  |
| 409        | Documents. Buyer has the Right to Terminate under § 25.1, on or before Association Documents Objection                     |
| 410        | Deadline (§ 3), based on any unsatisfactory provision in any of the Association Documents, in Buyer's sole                 |
| 411<br>412 | subjective discretion. Should Buyer receive the Association Documents after Association Documents Deadline                 |
| 413        | (§ 3), Buyer, at Buyer's option, has the Right to Terminate under § 25.1 by Buyer's Notice to Terminate received           |
|            | 10 // J / = -yp  |
|            |  |
| 4-9-12.    | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 5 of 15   |
|            | O.10   |
|            | an Arman March   |

Buyer(s) Initials\_ Seller(s) Initials by Seller on or before ten days after Buyer's receipt of the Association Documents. If Buyer does not receive the Association Documents, or if Buyer's Notice to Terminate would otherwise be required to be received by Seller after **Closing Date** (§ 3), Buyer's Notice to Terminate shall be received by Seller on or before **Closing** (§ 12.3). If Seller does not receive Buyer's Notice to Terminate within such time, Buyer accepts the provisions of the Association Documents as satisfactory, and Buyer waives any Right to Terminate under this provision, notwithstanding the provisions of § 8.5.

#### 8. RECORD TITLE AND OFF—RECORD TITLE.

- 8.1. Record Title. Buyer has the right to review and object to any of the Title Documents (Right to Object to Title, Resolution) as set forth in § 8.3. Buyer's objection may be based on any unsatisfactory form or content of Title Commitment, notwithstanding § 13, or any other unsatisfactory title condition, in Buyer's sole subjective discretion. If Buyer objects to any of the Title Documents, Buyer shall cause Seller to receive Buyer's Notice to Terminate or Notice of Title Objection on or before Record Title Objection Deadline (§ 3). If Title Documents are not received by Buyer, on or before the Record Title Deadline (§ 3), or if there is an endorsement to the Title Commitment that adds a new Exception to title, a copy of the new Exception to title and the modified Title Commitment shall be delivered to Buyer. Buyer shall cause Seller to receive Buyer's Notice to Terminate or Notice of Title Objection on or before ten days after receipt by Buyer of the following documents: (1) any required Title Document not timely received by Buyer, (2) any change to the Title Documents, or (3) endorsement to the Title Commitment. If Seller receives Buyer's Notice to Terminate or Notice of Title Objection, pursuant to this § 8.1 (Record Title), any title objection by Buyer and this Contract shall be governed by the provisions set forth in § 8.3 (Right to Object to Title, Resolution). If Seller does not receive Buyer's Notice to Terminate or Notice of Title Objection by the applicable deadline specified above, Buyer accepts the condition of title as disclosed by the Title Documents as satisfactory.
- 8.2. Off—Record Title. Seller shall deliver to Buyer, on or before Off—Record Title Deadline (§ 3), true copies of all existing surveys in Seller's possession pertaining to the Property and shall disclose to Buyer all easements, liens (including, without limitation, governmental improvements approved, but not yet installed) or other title matters (including, without limitation, rights of first refusal and options) not shown by public records, of which Seller has actual knowledge (Off-Record Matters). Buyer has the right to inspect the Property to investigate if any third party has any right in the Property not shown by public records (such as an unrecorded easement, unrecorded lease, boundary line discrepancy or water rights). Buyer's Notice to Terminate or Notice of Title Objection of any unsatisfactory condition (whether disclosed by Seller or revealed by such inspection, notwithstanding § 13), in Buyer's sole subjective discretion, shall be received by Seller on or before Off—Record Title Objection Deadline (§ 3). If Seller receives Buyer's Notice to Terminate or Notice of Title Objection pursuant to this § 8.2 (Off—Record Title), any title objection by Buyer and this Contract shall be governed by the provisions set forth in § 8.3 (Right to Object to Title, Resolution). If Seller does not receive Buyer's Notice to Terminate or Notice of Title Objection on or before Off—Record Title Objection Deadline (§ 3), Buyer accepts title subject to such rights, if any, of third parties of which Buyer has actual knowledge. Unless disclosed in writing, Seller represents and warrants that there are no Off-Record Matters.
- **8.3.** Right to Object to Title, Resolution. Buyer's Right to Object to Title shall include, but not be limited to those matters set forth in §§ 8.1 (Record Title), 8.2 (Off—Record Title) and 13 (Transfer of Title), in Buyer's sole subjective discretion (collectively, Right to Object to Title). If Buyer objects to any title matter, on or before the applicable deadline, Buyer shall have the option to either (1) object to the condition of title, or (2) terminate this Contract.
- **8.3.1.** Title Resolution. If Seller receives Buyer's Notice of Title Objection, as provided in § 8.1 (Record Title) or § 8.2 (Off—Record Title), on or before the applicable deadline, and if Buyer and Seller have not agreed to a written settlement thereof on or before **Title Resolution Deadline** (§ 3), this Contract shall terminate on the expiration of **Title Resolution Deadline** (§ 3) unless Seller receives Buyer's written withdrawal of Buyer's Notice of Title Objection, (i.e., Buyer's written notice to waive objection to such items and waives the right to terminate for that reason), on or before expiration of **Title Resolution Deadline** (§ 3).
- **8.3.2.** Right to Terminate Title Objection. Buyer has the Right to Terminate under § 25.1, on or before the applicable deadline, based on any unsatisfactory title matter, in Buyer's sole subjective discretion.
- 8.4. Special Taxing Districts. SPECIAL TAXING DISTRICTS MAY BE SUBJECT TO GENERAL OBLIGATION INDEBTEDNESS THAT IS PAID BY REVENUES PRODUCED FROM ANNUAL TAX LEVIES ON THE TAXABLE PROPERTY WITHIN SUCH DISTRICTS. PROPERTY OWNERS IN SUCH DISTRICTS MAY BE PLACED AT RISK FOR INCREASED MILL LEVIES AND TAX TO SUPPORT THE SERVICING OF SUCH DEBT WHERE CIRCUMSTANCES ARISE RESULTING IN THE INABILITY OF SUCH A DISTRICT TO DISCHARGE SUCH INDEBTEDNESS WITHOUT SUCH AN INCREASE IN MILL LEVIES. BUYERS SHOULD INVESTIGATE THE SPECIAL TAXING DISTRICTS IN WHICH THE PROPERTY IS LOCATED BY CONTACTING THE COUNTY TREASURER, BY REVIEWING THE CERTIFICATE OF TAXES DUE FOR THE PROPERTY, AND BY OBTAINING FURTHER INFORMATION FROM THE BOARD OF COUNTY COMMISSIONERS, THE COUNTY CLERK AND RECORDER, OR THE COUNTY ASSESSOR.

Buyer has the Right to Terminate under § 25.1, on or before Off—Record Title Objection Deadline (§ 3), based on any unsatisfactory effect of the Property being located within a special taxing district, in Buyer's sole

Page 6 of 15

|                   | 1 | $-\mathcal{W}_{h}$       |
|-------------------|---|--------------------------|
| Buyer(s) Initials |   | Seller(s) Initials // // |
|                   | • |                          |

CONTRACT TO BUY AND SELL REAL ESTATE - Land

CTMeContracts.com - ©2013 CTM Software Corp.

- 8.5. Right of First Refusal or Contract Approval. If there is a right of first refusal on the Property, or a right to approve this Contract, Seller shall promptly submit this Contract according to the terms and conditions of such right. If the holder of the right of first refusal exercises such right or the holder of a right to approve disapproves this Contract, this Contract shall terminate. If the right of first refusal is waived explicitly or expires, or the Contract is approved, this Contract shall remain in full force and effect. Seller shall promptly notify Buyer in writing of the foregoing. If expiration or waiver of the right of first refusal or Contract approval has not occurred on or before Right of First Refusal Deadline (§ 3), this Contract shall then terminate.
- 8.6. Title Advisory. The Title Documents affect the title, ownership and use of the Property and should be reviewed carefully. Additionally, other matters not reflected in the Title Documents may affect the title, ownership and use of the Property, including, without limitation, boundary lines and encroachments, area, zoning, unrecorded easements and claims of easements, leases and other unrecorded agreements, and various laws and governmental regulations concerning land use, development and environmental matters. The surface estate may be owned separately from the underlying mineral estate, and transfer of the surface estate does not necessarily include transfer of the mineral rights or water rights. Third parties may hold interests in oil, gas, other minerals, geothermal energy or water on or under the Property, which interests may give them rights to enter and use the Property. Such matters may be excluded from or not covered by the title insurance policy. Buyer is advised to timely consult legal counsel with respect to all such matters as there are strict time limits provided in this Contract [e.g., Record Title Objection Deadline (§ 3)].

#### 9. CURRENT SURVEY REVIEW.

| o. Outstant Outstant stanton.  |
|--|
| 9.1. Current Survey Conditions. If the box in § 9.1.1 or § 9.1.2 is checked, Buyer, the issuer of the Title  |
| Commitment or the provider of the opinion of title if an Abstract, and   |
| shall receive a Current Survey, i.e., Improvement Location Certificate, Improvement Survey Plat or other form of   |
| survey set forth in § 9.1.2 (collectively, Current Survey), on or before Current Survey Deadline (§ 3). The Current  |
| Survey shall be certified by the surveyor to all those who are to receive the Current Survey.  |
| 9.1.1. Improvement Location Certificate. If the box in this § 9.1.1 is checked, Seller Buyer   |
| shall order or provide, and pay, on or before Closing, the cost of an Improvement Location Certificate.  |
| 9.1.2. Other Survey. If the box in this § 9.1.2 is checked, a Current Survey, other than an  |
| Improvement Location Certificate, shall be an Improvement Survey Plat Improvem |
| parties agree that payment of the cost of the Current Survey and obligation to order or provide the Current Survey   |
| shall be as follows:   |
| 9.2. Current Objection. Buyer has the right to review and object to the Current Survey. Buyer has the  |
| Right to Terminate under § 25.1, on or before the Current Survey Objection Deadline (§ 3), if the Current Survey   |
| is not timely received by Buyer or based on any unsatisfactory matter with the Current Survey, notwithstanding §   |
| 8.2 or § 13.   |

# DISCLOSURE, INSPECTION AND DUE DILIGENCE

- 10. PROPERTY DISCLOSURE, INSPECTION, INDEMNITY, INSURABILITY, DUE DILIGENCE AND SOURCE OF WATER.
- 10.1. Seller's Property Disclosure. On or before Seller's Property Disclosure Deadline (§ 3), Seller agrees to deliver to Buyer the most current version of the applicable Colorado Real Estate Commission's Seller's Property Disclosure form completed by Seller to Seller's actual knowledge, current as of the date of this Contract.
- 10.2. Inspection Objection. Unless otherwise provided in this Contract, Buyer acknowledges that Seller is conveying the Property to Buyer in an "as is" condition, "where is" and "with all faults". Seller shall disclose to Buyer, in writing, any latent defects actually known by Seller. Buyer, acting in good faith, has the right to have inspections (by one or more third parties, personally or both) of the Property and Inclusions (Inspection), at Buyer's expense. If (1) the physical condition of the Property, including but not limited to, the roof, walls, structural integrity of the Property, the electrical, plumbing, HVAC and other mechanical systems of the Property, (2) the physical condition of the Inclusions, (3) service to the Property (including utilities and communication services), systems and components of the Property, e.g. heating and plumbing, (4) any proposed or existing transportation project, road, street or highway, or (5) any other activity, odor or noise (whether on or off the Property) and its effect or expected effect on the Property or its occupants is unsatisfactory, in Buyer's sole subjective discretion, Buyer shall, on or before Inspection Objection Deadline (§ 3):

|      | 5/1<br>572<br>573<br>574<br>575<br>576<br>577 | Buyer shall, on or before Inspection Objection Deadline (§ 3):  10.2.1. Notice to Terminate. Notify Seller in writing that this Contract is terminated; or  10.2.2. Inspection Objection. Deliver to Seller a written description of any unsatisfactory physical condition that Buyer requires Seller to correct. |
|------|---|---|
|      | 578   | Buyer has the Right to Terminate under § 25.1, on or before Inspection Objection Deadline (§ 3) if the  |
|      | 579   | Property or Inclusions are unsatisfactory, in Buyer's sole subjective discretion.   |
| CBS4 | -9-12.<br>r(s) Initi                          | als Seller(s) Initials CTMeContracts.com - ©2013 CTM Software Corp.   |
|      | •   |   |
|      |   |   |

599

600

618 619

609

610

626

635

636

637

> 646 647

> 648

649 650

662

663

- 10.3. Inspection Resolution. If an Inspection Objection is received by Seller, on or before Inspection Objection Deadline (§ 3) and if Buyer and Seller have not agreed in writing to a settlement thereof on or before Inspection Resolution Deadline (§ 3), this Contract shall terminate on Inspection Resolution Deadline (§ 3). unless Seller receives Buyer's written withdrawal of the Inspection Objection before such termination, i.e., on or before expiration of Inspection Resolution Deadline (§ 3).
- 10.4. Damage, Liens and Indemnity. Buyer, except as otherwise provided in this Contract or other written agreement between the parties, is responsible for payment for all inspections, tests, surveys, engineering reports, or other reports performed at Buyer's request (Work) and shall pay for any damage that occurs to the Property and Inclusions as a result of such Work. Buyer shall not permit claims or liens of any kind against the Property for Work performed on the Property at Buyer's request. Buyer agrees to indemnify, protect and hold Seller harmless from and against any liability, damage, cost or expense incurred by Seller and caused by any such Work, claim, or lien. This indemnity includes Seller's right to recover all costs and expenses incurred by Seller to defend against any such liability, damage, cost or expense, or to enforce this section, including Seller's reasonable attorney fees, legal fees and expenses. The provisions of this section shall survive the termination of this Contract. This § 10.4 does not apply to items performed pursuant to an Inspection Resolution.
- 10.5. Insurability. Buyer has the right to review and object to the availability, terms and conditions of and premium for property insurance (Property Insurance). Buyer has the Right to Terminate under § 25.1, on or before Property Insurance Objection Deadline (§ 3), based on any unsatisfactory provision of the Property Insurance, in Buyer's sole subjective discretion.
- 10.6. Due Diligence Documents. Seller agrees to deliver copies of the following documents and information pertaining to the Property (Due Diligence Documents) to Buyer on or before Due Diligence **Documents Delivery Deadline(§ 3):** 
  - 10.6.1. All contracts relating to the operation, maintenance and management of the Property:
  - 10.6.2. Property tax bills for the last 1 years;
- 10.6.3. As-built construction plans to the Property and the tenant improvements, including architectural, electrical, mechanical, and structural systems; engineering reports; and permanent Certificates of Occupancy, to the extent now available:
  - 10.6.4. A list of all Inclusions to be conveyed to Buyer;
  - **10.6.5.** Operating statements for the past **0** years;
  - 10.6.6. A rent roll accurate and correct to the date of this Contract;
- 10.6.7. All current leases, including any amendments or other occupancy agreements, pertaining to the Property (Leases);
- 10.6.8. A schedule of any tenant improvement work Seller is obligated to complete but has not yet completed and capital improvement work either scheduled or in process on the date of this Contract;
- 10.6.9. All insurance policies pertaining to the Property and copies of any claims which have been made for the past 0 years;
- 10.6.10. Soils reports, Surveys and engineering reports or data pertaining to the Property (if not delivered earlier under § 8.2);
- 10.6.11. Any and all existing documentation and reports regarding Phase I and II environmental reports, letters, test results, advisories, and similar documents respective to the existence or nonexistence of asbestos, PCB transformers, or other toxic hazardous or contaminated substances, and/or underground storage tanks and/or radon gas. If no reports are in Seller's possession or known to Seller, Seller warrants that no such reports are in Seller's possession or known to Seller:
- 10.6.12. Any Americans with Disabilities Act reports, studies or surveys concerning the compliance of the Property with said Act;
- 10.6.13. All permits, licenses and other building or use authorizations issued by any governmental authority with jurisdiction over the Property and written notice of any violation of any such permits, licenses or use authorizations, if any; and
  - 10.6.14. Other Documents:

NONE

- 10.7. Due Diligence Documents Conditions. Buyer has the right to review and object to Due Diligence Documents, zoning and any use restrictions imposed by any governmental agency with jurisdiction over the Property (Zoning), in Buyer's sole subjective discretion, and has the right to object if Seller fails to deliver to Buyer all Due Diligence Documents. Buyer shall also have the unilateral right to waive any condition herein.
- 10.7.1. Due Diligence Documents Objection. Buyer has the Right to Terminate under § 25.1, on or before Due Diligence Documents Objection Deadline (§ 3), based on any unsatisfactory matter with the Due Diligence Documents, in Buyer's sole subjective discretion. If all Due Diligence Documents under § 10.6 are not received by Buyer on or before Due Diligence Documents Delivery Deadline (§ 3), then Buyer has the Right to Terminate under § 25.1 on or before the earlier of ten days after **Due Diligence Documents Objection** Deadline (§ 3) or Closing.
- 10.7.2. Zoning. Buyer has the Right to Terminate under § 25.1, on or before Due Diligence Documents Objection Deadline (§ 3), based on any unsatisfactory zoning, in Buyer's sole subjective discretion.

| CBS4-9-12. CO     | NTRACT TO BUY AND SELL REAL ESTAT | E - Land | Page 8 of 1: | 5  |
|-------------------|-----------------------------------|----------|--------------|----|
|                   |                                   |          | -71          | ), |
| Buyer(s) Initials |                                   | Seller(s | ) Initials   | 1  |

| 664        |  |
|------------|--|
| 665        | 10.7.3. Source of Potable Water (Residential Land and Residential Improvements Only).  |
| 666<br>667 |  |
| 668        | Addendum disclosing the source of potable water for the Property. Buyer Li Does Li Does Not acknowledge  |
| 669        |  |
| 670        |  |
| 671<br>672 |  |
| 673        |  |
| 674        | DETERMINE THE LONG-TERM SUFFICIENCY OF THE PROVIDER'S WATER SUPPLIES.  |
| 675<br>676 | 10.9 Due Diligence Environmental ADA Buyer shall have the right to obtain environmental  |
| 677        |  |
| 678        |  |
| 679        | Environmental Cita Accessment (consultant with ACTM E4507 05 standard practices for Environmental Cita   |
| 680<br>681 |  |
| 682        |  |
| 683        | addition, buyer may also conduct an evaluation whether the Property compiles with the Americans with   |
| 684        |  |
| 685<br>686 | manamy agreement to minimize the minimized the mini |
| 687        | ii any.  |
| 688        | If Buyer's Phase I Environmental Site Assessment recommends a Phase II Environmental Site  |
| 689        |  |
| 690<br>691 |  |
| 692        |  |
| 693        | Buyer shall have the Right to Terminate under § 25.1, on or before Environmental Inspection Objection  |
| 694        | Pardline (8.2) or if applicable the Extended Environmental Increation Objection Deadline, based on any   |
| 695<br>696 | unsatisfactory results of Environmental Inspection, in Buyer's sole subjective discretion.  Buyer shall have   |
| 697        | the Right to Terminate under § 25.1, on or before ADA Evaluation Objection Deadline (§ 3), based on any  |
| 698        | uncertisfectory ADA Evaluation in Puyer's cale subjective discretion   |
| 699<br>700 |  |
| 701        | the Leases to be assigned to the Buyer at the time of Closing contain any rent concessions, rent reductions or   |
| 702        | rent abatements except as disclosed in the Lease or other writing received by Buyer. Seller shall not amend, alter,  |
| 703        | modify, extend or cancel any of the Leases nor shall Seller enter into any new leases affecting the Property   |
| 704<br>705 |  |
| 706        | without the prior written consent of Buyer, which consent shall not be unreasonably withheld or delayed.   |
| 707        | AL TENANT FORODEL OF TENANTS   |
| 708<br>709 | 11. TENANT ESTOPPEL STATEMENTS.  |
| 710        | 11.1. Tenant Estoppel Statements Conditions. Buyer has the right to review and object to any   |
| 711        | Estoppel Statements. Seller shall obtain and deliver to Buyer on or before <b>Tenant Estoppel Statements</b>   |
| 712        | Deadline (§ 3), statements in a form and substance reasonably acceptable to Buyer, from each occupant or   |
| 713<br>714 | tenant at the Property (Estoppel Statement) attached to a copy of such occupant's or tenant's lease and any  |
| 715        | amendments (Lease) stating:  |
| 716        | 11.1.1. The commencement date of the Lease and scheduled termination date of the Lease;  |
| 717<br>718 | 11.1.2. That said Lease is in full force and effect and that there have been no subsequent   |
| 719        | modifications or amendments;   |
| 720        | 11.1.3. The amount of any advance rentals paid, rent concessions given, and deposits paid to   |
| 721        | Seller;  |
| 722<br>723 | 11.1.4. The amount of monthly (or other applicable period) rental paid to Seller;  |
| 724        | 11.1.5. That there is no default under the terms of said Lease by landlord or occupant; and  |
| 725        | 11.1.6. That the Lease to which the Estoppel is attached is a true, correct and complete copy of   |
| 726<br>727 | the Lease demising the premises it describes.  |
| 727<br>728 | 11.2. Tenant Estoppel Statements Objection Buyer has the Right to Terminate under § 25.1, on or  |
| 729        | before Tenant Estoppel Statements Objection Deadline (§ 3), based on any unsatisfactory Estoppel   |
| 730        | Statement, in Buyer's sole subjective discretion or if Seller fails to deliver the Estoppel Statements on or before  |
| 731<br>732 | Tenant Estoppel Statements Deadline (§ 3). Buyer shall also have the unilateral right to waive any   |
| 733        | unsatisfactory Estoppel Statement.   |
| 734        | and another the control of the contr |
| 735<br>736 | CLOSING PROVISIONS   |
| 737        | CLUSING PROVISIONS   |
| 738        |  |
| 739<br>740 | 12. CLOSING DOCUMENTS, INSTRUCTIONS AND CLOSING.   |
| 740<br>741 | 12.1. Closing Documents and Closing Information. Seller and Buyer shall cooperate with the Closing   |
| 742        | Company to enable the Closing Company to prepare and deliver documents required for Closing to Buyer and   |
| 743        | Seller and their designees. If Buyer is obtaining a new loan to purchase the Property, Buyer acknowledges  |
| 744<br>745 | Buyer's lender shall be required to provide the Closing Company in a timely manner all required loan documents   |
| 745        | and financial information concerning Buyer's new loan. Buyer and Seller will furnish any additional information and  |
| 747        | documents required by Closing Company that will be necessary to complete this transaction. Buyer and Seller  |
| CBS4-9-12. | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 9 of 15   |
|            | · ·  |

Buyer(s) Initials

ZJF

G.11. 7.X F.22.1.

Mus

| Buyer(s) Init     | ials Seller(s) Initials  |
|-------------------|--|
|                   | ma (   |
| 830<br>CBS4-9-12. | 16. PRORATIONS. The following shall be prorated to Closing Date (§ 3), except as otherwise provided:  CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 10 of 15                          |
| 828<br>829        |  |
| 826<br>827        | 15.7. Sales and Use Tax. Any sales and use tax that may accrue because of this transaction shall be paid when due by ☑ Buyer ☐ Seller ☐ One-Half by Buyer and One-Half by Seller ☐ None. |
| 824<br>825        | 15.7. Sales and Use Tax. Any sales and use tax that may accrue because of this transaction   |
| 823               | and must be paid at Closing by $\square$ Buyer $\square$ Seller $\square$ One-Half by Buyer and One-Half by Seller $\square$ None.   |
| 821<br>822        | \$ <b>n/a</b> for □ Augmentation Membership □ Small Domestic Water Company □ <b>n/a</b>  |
| 820               | \$ <b>n/a</b> for □ Water Stock/ Certificates □ Water District   |
| 818<br>819        | Contract, do not exceed:   |
| 816<br>817        | amount of <u>TBD</u> % of the Purchase Price or \$  15.6. Water Transfer Fees. The Water Transfer Fees can change. The fees, as of the date of this                                      |
| 815               | whether one or more, is for the following association(s): <b>CANNON VIEW HOA</b> in the total  |
| 813<br>814        | Closing by Buyer Seller One-Half by Buyer and One-Half by Seller None. The Private Transfer fee,   |
| 811<br>812        | payable at Closing, such as community association fees, developer fees and foundation fees, shall be paid at   |
| 810               | 15.5. Private Transfer Fee. Private transfer fees and other fees due to a transfer of the Property,  |
| 808<br>809        | shall be paid at Closing by Buyer Seller One-Half by Buyer and One-Half by Seller Ono.   |
| 806<br>807        | paid by ⊠ Buyer □ Seller □ One-Half by Buyer and One-Half by Seller □ None.  15.4. Local Transfer Tax. □ The Local Transfer Tax of N/A % of the Purchase Price                           |
| 805               | ownership record transfer fees regardless of name or title of such fee (Association's Record Change Fee) shall be  |
| 803<br>804        | One-Half by Seller None. Any record change fee assessed by the Association including, but not limited to,  |
| 802               | statement of assessments (Status Letter) shall be paid by Buyer Seller One-Half by Buyer and   |
| 800<br>801        | 15.3. Status Letter and Record Change Fees. Any fees incident to the issuance of Association's   |
| 798<br>799        | ☑ Buyer ☐ Seller ☐ One-Half by Buyer and One-Half by Seller ☐ Other <i>N/A</i>   |
| 797               | 15.2. Closing Services Fee. The fee for real estate closing services shall be paid at Closing by   ☑ Buyer ☐ Seller ☐ One-Half by Buyer and One-Half by Seller                           |
| 795<br>796        | all other items required to be paid at Closing, except as otherwise provided herein.   |
| 793<br>794        | 15.1. Closing Costs. Buyer and Seller shall pay, in Good Funds, their respective closing costs and   |
| 792               | 15. CLOSING COSTS, CLOSING FEE, ASSOCIATION FEES AND TAXES.  |
| 790<br>791        | eleaning from the procedure of this transaction of from any other source.  |
| 788<br>789        | 14. PAYMENT OF ENCUMBRANCES. Any encumbrance required to be paid shall be paid at or before Closing from the proceeds of this transaction or from any other source.                      |
| 787               | 14 DAVMENT OF ENGLINDRANGES Any oncumbrance required to be neid shall be neid at an hefere   |
| 785<br>786        | 13.5. Other with the Jefferson County Clerk and Recorder's Office  |
| 783<br>784        | of Canyon View per document recorded at Reception No. 2007030761,  |
| 782               | the associated Declarations of Covenants, Conditions and Restrictions  |
| 780<br>781        | 13.4. Inclusion of the Property within any special taxing district, and inclusion of the property in the Canyon View Owner's Association, and  |
| 779               | 40 A leaking of the Department in a second district and  |
| 777<br>778        | ,  |
| 775<br>776        | 13.3. Those specifically described rights of third parties not shown by the public records of which  |
| 774               | 13.2. Distribution utility easements (including cable TV).   |
| 772<br>773        |  |
| 770<br>771        | 12.1 Thora energific Eventions described by reference to recorded documents as reflected in the  |
| 769               | liens, including any governmental liens for special improvements installed as of the date of Buyer's signature   |
| 767<br>768        | the general taxes for the year of Closing. Except as provided herein, title shall be conveyed free and clear of all  |
| 766               | SPECIAL deed to Buyer, at Closing, conveying the Property free and clear of all taxes except   |
| 764<br>765        | Discountible of the terms and explicit to terior to payment and coloning as required referent and compilated by  |
| 762<br>763        |  |
| 761               | companies).  |
| 759<br>760        | service vary between different settlement service providers (e.g., attorneys, lenders, inspectors and title  |
| 757<br>758        |  |
| 756               | shall be so designated by TUE PAPTIES  |
| 754<br>755        | date specified as the Closing Date (§ 3) or by mutual agreement at an earlier date. The hour and place of Closing  |
| 752<br>753        | 12.3. Closing, Delivery of dead from Saller to Buyer shall be at closing (Closing). Closing shall be on the  |
| 750<br>751        |  |
| 748<br>749        | shall sign and complete all customary or reasonably required documents at or before closing.   |
|                   |  |

CTMeContracts.com - ©2013 CTM Software Corp.

| 831  | Taxos: 1 crsonal property taxos, if any, special taxing district accommons, if any, and general  |  |
|--|--|--|
| 832<br>833   |  |  |
| 834  |  |  |
| 835  |  |  |
| 836<br>837   | - m/a  |  |
| 838  |  |  |
| 839  | transfer or credit to Buyer the cognitive deposite for all Legges assigned, or any remainder after lawful deductions   |  |
| 840  |  |  |
| 841<br>842   |  |  |
| 843  | bayor an Education of Check at Closing and Dayor Shan assume Selicit's Obligations and Closic Statistics   |  |
| 844  | 9  |  |
| 845<br>846   | · · · · · · · · · · · · · · · · · · ·  |  |
| 847  | recognition recognitions for deferred maintenance by the respondition shall not be ordated to concrete except as   |  |
| 848  | may be otherwise provided by the Governing Documents. Buyer acknowledges that Buyer may be obligated to pay  |  |
| 849<br>850   |  |  |
| 851  | electing bate (3 c) by the Association shall be the obligation of Elbayer - Except however, any  |  |
| 852  | special assessment by the Association for improvements that have been installed as of the date of Buyer's  |  |
| 853  |  |  |
| 854<br>855   |  |  |
| 856  |  |  |
| 857  | Such assessments are subject to change as provided in the Governing Documents. Seller agrees to promptly   |  |
| 858<br>859   | request the Association to deliver to Ruyer before Closing Date (\$3) a current Status Letter  |  |
| 860  | 40.4 Other Broadface Miles I and All Miles I are a fine and an analysis of the second and an analysis of the second and the se |  |
| 861  |  |  |
| 862  |  |  |
| 863<br>864   |  |  |
| 865  | · ·  |  |
| 866  | NONE   |  |
| 867<br>868   |  |  |
| 869  |  |  |
| 870  | Because in Data (6.0) and Decease in Time (6.0) and I was a raise in data and  |  |
| 871<br>872   |  |  |
| 872<br>873   |  |  |
| 874  | GENERAL PROVISIONS   |  |
| 875  |  |  |
| 876<br>877   |  |  |
| 878  | 18.1. Day. As used in this Contract, the term "day" shall mean the entire day ending at 11:59 p.m., United   |  |
| 879<br>880   | outloo Mountain Time (Clandard of Bayingth Cavings as applicable).   |  |
| 881  | 18.2. Computation of Period of Days, Deadline. In computing a period of days, when the ending date   |  |
| 882  |  |  |
| 883<br>884   | rails on a Saturday, Suriday of receita of Colorado State Holiday (Holiday), Such deadline 23 Shah 🗀 Shah Not  |  |
| 885  |  |  |
| 886  |  |  |
| 887<br>888   |  |  |
| 889  | 19. CAUSES OF LOSS, INSURANCE; DAMAGE TO INCLUSIONS AND SERVICES; CONDEMNATION;  |  |
| 890  |  |  |
| 891<br>892   | delivered in the condition existing as of the date of this contract, ordinary wear and tear excepted.  |  |
| 893  |  |  |
| 894  | perils or causes of loss prior to Closing in an amount of not more than ten percent of the total Purchase Price  |  |
| 895<br>896   |  |  |
|  |  |  |
| 897  |  |  |
| 898  | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such   |  |
| 898<br>899   | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such   |  |
| 898  | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such   |  |
| 898<br>899<br>900<br>901<br>902  | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount  |  |
| 898<br>899<br>900<br>901<br>902<br>903   | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the  |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904  | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the  |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905   | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.   |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>905  | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and  |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>906<br>907   | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and  |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>905  | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and  |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>907<br>908<br>909<br>910<br>911                      | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and communication services), system, component or fixture of the Property (collectively Service), e.g., heating or   |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>907<br>908<br>909<br>910<br>911                      | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and communication services), system, component or fixture of the Property (collectively Service), e.g., heating or plumbing, fail or be damaged between the date of this Contract and Closing or possession, whichever shall be earlier, then Seller shall be liable for the repair or replacement of such Inclusion or Service with a unit of similar   |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>907<br>908<br>909<br>910<br>911                      | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and communication services), system, component or fixture of the Property (collectively Service), e.g., heating or plumbing, fail or be damaged between the date of this Contract and Closing or possession, whichever shall be earlier, then Seller shall be liable for the repair or replacement of such Inclusion or Service with a unit of similar size, age and quality, or an equivalent credit, but only to the extent that the maintenance or replacement of such  |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>906<br>907<br>908<br>909<br>910<br>911<br>912        | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and communication services), system, component or fixture of the Property (collectively Service), e.g., heating or plumbing, fail or be damaged between the date of this Contract and Closing or possession, whichever shall be earlier, then Seller shall be liable for the repair or replacement of such Inclusion or Service with a unit of similar size, age and quality, or an equivalent credit, but only to the extent that the maintenance or replacement of such Inclusion or Service is not the responsibility of the Association, if any, less any insurance proceeds received by   |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>906<br>907<br>908<br>909<br>910<br>911<br>912<br>913 | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and communication services), system, component or fixture of the Property (collectively Service), e.g., heating or plumbing, fail or be damaged between the date of this Contract and Closing or possession, whichever shall be earlier, then Seller shall be liable for the repair or replacement of such Inclusion or Service with a unit of similar size, age and quality, or an equivalent credit, but only to the extent that the maintenance or replacement of such Inclusion or Service is not the responsibility of the Association, if any, less any insurance proceeds received by   |  |
| 898<br>899<br>900<br>901<br>902<br>903<br>904<br>905<br>906<br>907<br>908<br>909<br>910<br>911<br>912<br>913 | Date (§ 3) or if the damage exceeds such sum. Should Buyer elect to carry out this Contract despite such Property Damage, Buyer shall be entitled to a credit at Closing for all insurance proceeds that were received by Seller (but not the Association, if any) resulting from such damage to the Property and Inclusions, plus the amount of any deductible provided for in such insurance policy. Such credit will not exceed the Purchase Price. In the event Seller has not received such insurance proceeds prior to Closing, the parties may agree to extend the Closing Date (§ 3) or, at the option of Buyer, Seller shall assign such proceeds at Closing, plus credit Buyer the amount of any deductible provided for in such insurance policy, but not to exceed the total Purchase Price.  19.2. Damage, Inclusions and Services. Should any Inclusion or service (including utilities and communication services), system, component or fixture of the Property (collectively Service), e.g., heating or plumbing, fail or be damaged between the date of this Contract and Closing or possession, whichever shall be earlier, then Seller shall be liable for the repair or replacement of such Inclusion or Service with a unit of similar size, age and quality, or an equivalent credit, but only to the extent that the maintenance or replacement of such Inclusion or Service is not the responsibility of the Association, if any, less any insurance proceeds received by   |  |

 ${\bf CTMe Contracts.com - @2013\ CTM\ Software\ Corp.}$ 

- 19.3. Condemnation. In the event Seller receives actual notice prior to Closing that a pending condemnation action may result in a taking of all or part of the Property or Inclusions, Seller shall promptly notify Buyer, in writing, of such condemnation action. Buyer has the Right to Terminate under § 25.1, on or before Closing Date (§ 3), based on such condemnation action, in Buyer's sole subjective discretion. Should Buyer elect to consummate this Contract despite such diminution of value to the Property and Inclusions. Buyer shall be entitled to a credit at Closing for all condemnation proceeds awarded to Seller for the diminution in the value of the Property or Inclusions but such credit shall not include relocation benefits or expenses, or exceed the Purchase Price.
- 19.4. Walk—Through and Verification of Condition. Buyer, upon reasonable notice, has the right to walk through the Property prior to Closing to verify that the physical condition of the Property and Inclusions complies with this Contract.
- 19.5. Risk of Loss Growing Crops. The risk of loss for damage to growing crops by fire or other casualty shall be borne by the party entitled to the growing crops as provided in § 2.5.5 and such party shall be entitled to such insurance proceeds or benefits for the growing crops
- RECOMMENDATION OF LEGAL AND TAX COUNSEL. By signing this Contract, Buyer and Seller acknowledge that the respective broker has advised that this Contract has important legal consequences and has recommended the examination of title and consultation with legal and tax or other counsel before signing this Contract.
- 21. TIME OF ESSENCE, DEFAULT AND REMEDIES. Time is of the essence hereof. If any note or check received as Earnest Money hereunder or any other payment due hereunder is not paid, honored or tendered when due, or if any obligation hereunder is not performed or waived as herein provided, the non-defaulting party has the following remedies:
  - 21.1. If Buyer is in Default:

916

917 918

919

920

921 922

923

924

925

926 927

928

929 930

931

932

933 934

935

936

937

938 939

940

941

942 943 944

945

947

948

949 950 951

952

953 954

955

956

957

958 959

960

961 962

963

964

965 966

967

968

969 970

971

972

973 974

975

976

977 978 979

980

981 982

983 984

985

987

988

989 990

- 21.1.1. Specific Performance. Seller may elect to treat this Contract as canceled, in which case all Earnest Money (whether or not paid by Buyer) shall be paid to Seller and retained by Seller; and Seller may recover such damages as may be proper; or Seller may elect to treat this Contract as being in full force and effect and Seller has the right to specific performance or damages, or both.
- 21.1.2. Liquidated Damages, Applicable. This § 21.1.2 shall apply unless the box in § 21.1.1. is checked. All Earnest Money (whether or not paid by Buyer) shall be paid to Seller, and retained by Seller. Both parties shall thereafter be released from all obligations hereunder. It is agreed that the Earnest Money specified in § 4.1 is LIQUIDATED DAMAGES, and not a penalty, which amount the parties agree is fair and reasonable and (except as provided in §§ 10.4, 22, 23 and 24), said payment of Earnest Money shall be SELLER'S ONLY REMEDY for Buyer's failure to perform the obligations of this Contract. Seller expressly waives the remedies of specific performance and additional damages.
- 21.2. If Seller is in Default: Buyer may elect to treat this Contract as canceled, in which case all Earnest Money received hereunder shall be returned and Buyer may recover such damages as may be proper, or Buyer may elect to treat this Contract as being in full force and effect and Buyer has the right to specific performance or damages, or both.
- LEGAL FEES, COST AND EXPENSES. Anything to the contrary herein notwithstanding, in the event of any arbitration or litigation relating to this Contract, prior to or after Closing Date (§ 3), the arbitrator or court shall award to the prevailing party all reasonable costs and expenses, including attorney fees, legal fees and expenses.
- MEDIATION. If a dispute arises relating to this Contract, prior to or after Closing, and is not resolved, the parties shall first proceed in good faith to submit the matter to mediation. Mediation is a process in which the parties meet with an impartial person who helps to resolve the dispute informally and confidentially. Mediators cannot impose binding decisions. The parties to the dispute must agree, in writing, before any settlement is binding. The parties will jointly appoint an acceptable mediator and will share equally in the cost of such

| 991<br>992<br>993<br>994 | mediation. The mediation, unless otherwise agreed, shall terminate in the event the entire dispute is not resolved within thirty days of the date written notice requesting mediation is delivered by one party to the other at the party's last known address. This section shall not alter any date in this Contract, unless otherwise agreed. |
|--------------------------|--|
| 995<br>996<br>997<br>998 | 24. EARNEST MONEY DISPUTE. Except as otherwise provided herein, Earnest Money Holder shall release the Earnest Money as directed by written mutual instructions, signed by both Buyer and Seller. In the event of an   |
| CBS4-9-12.               | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 12 of 15  |
| Buyer(s) Init            | sialsSeller(s) Initials  |
|                          | CTMeContracts.com - ©2013 CTM Software Corp.   |
|                          |  |

controversy regarding the Earnest Money (notwithstanding any termination of this Contract), Earnest Money Holder shall not be required to take any action. Earnest Money Holder, at its sole subjective discretion, has several options: (1) await any proceeding, (2) interplead all parties and deposit Earnest Money into a court of competent jurisdiction and shall recover court costs and reasonable attorney and legal fees, or (3) provide notice to Buyer and Seller that unless Earnest Money Holder receives a copy of the Summons and Complaint or Claim (between Buyer and Seller) containing the case number of the lawsuit (Lawsuit) within one hundred twenty days of Earnest Money Holder's notice to the parties, Earnest Money Holder shall be authorized to return the Earnest Money to Buyer. In the event Earnest Money Holder does receive a copy of the Lawsuit, and has not interplead the monies at the time of any Order, Earnest Money Holder shall disburse the Earnest Money pursuant to the Order of the Court. The parties reaffirm the obligation of **Mediation** (§ 23).

#### 25. TERMINATION.

- **25.1. Right to Terminate.** If a party has a right to terminate, as provided in this Contract (Right to Terminate), the termination shall be effective upon the other party's receipt of a written notice to terminate (Notice to Terminate), provided such written notice was received on or before the applicable deadline specified in this Contract. If the Notice to Terminate is not received on or before the specified deadline, the party with the Right to Terminate shall have accepted the specified matter, document or condition as satisfactory and waived the Right to Terminate under such provision.
- **25.2. Effect of Termination.** In the event this Contract is terminated, all Earnest Money received hereunder shall be returned and the parties shall be relieved of all obligations hereunder, subject to §§ 10.4, 22, 23 and 24.
- 26. ENTIRE AGREEMENT, MODIFICATION, SURVIVAL. This Contract, its exhibits and specified addenda, constitute the entire agreement between the parties relating to the subject hereof, and any prior agreements pertaining thereto, whether oral or written, have been merged and integrated into this Contract. No subsequent modification of any of the terms of this Contract shall be valid, binding upon the parties, or enforceable unless made in writing and signed by the parties. Any right or obligation in this Contract that, by its terms, exists or is intended to be performed after termination or Closing shall survive the same.

### 27. NOTICE, DELIVERY, AND CHOICE OF LAW.

CONTRACT TO BUY AND SELL REAL ESTATE - Land

- 27.1. Physical Delivery. All notices must be in writing, except as provided in § 27.2. Any document, including a signed document or notice, from or on behalf of Seller, and delivered to Buyer shall be effective when physically received by Buyer, any signatory on behalf of Buyer, any named individual of Buyer, any representative of Buyer, or Brokerage Firm of Broker working with Buyer (except for delivery, after Closing, of the notice requesting mediation described in § 23) and except as provided in § 27.2. Any document, including a signed document or notice, from or on behalf of Buyer, and delivered to Seller shall be effective when physically received by Seller, any signatory on behalf of Seller, any named individual of Seller, any representative of Seller, or Brokerage Firm of Broker working with Seller (except for delivery, after Closing, of the notice requesting mediation described in § 23) and except as provided in § 27.2.
- 27.2. Electronic Delivery. As an alternative to physical delivery, any document, including any signed document or written notice, may be delivered in electronic form only by the following indicated methods: Facsimile E-mail Internet INO Electronic Delivery. If the box "No Electronic Delivery" is checked, this § 27.2 shall not be applicable and § 27.1 shall govern notice and delivery. Documents with original signatures shall be provided upon request of any party.
- **27.3.** Choice of Law. This Contract and all disputes arising hereunder shall be governed by and construed in accordance with the laws of the State of Colorado that would be applicable to Colorado residents who sign a contract in Colorado for property located in Colorado.
- 28. NOTICE OF ACCEPTANCE, COUNTERPARTS. This proposal shall expire unless accepted in writing, by Buyer and Seller, as evidenced by their signatures below, and the offering party receives notice of such acceptance pursuant to § 27 on or before Acceptance Deadline Date (§ 3) and Acceptance Deadline Time (§ 3). If accepted, this document shall become a contract between Seller and Buyer. A copy of this Contract may be executed by each party, separately, and when each party has executed a copy thereof, such copies taken together shall be deemed to be a full and complete contract between the parties.
- 29. GOOD FAITH. Buyer and Seller acknowledge that each party has an obligation to act in good faith, including but not limited to exercising the rights and obligations set forth in the provisions of Financing Conditions and Obligations (§ 5), Record Title and Off-Record Title (§ 8), Current Survey Review (§ 9) and Property Disclosure, Inspection, Indemnity, Insurability, Due Diligence, Buyer Disclosure and Source of Water (§ 10).

### ADDITIONAL PROVISIONS AND ATTACHMENTS

Page 13 of 15

| <u> </u>          | $\mathcal{M}_{\mathcal{A}}$ |
|-------------------|-----------------------------|
| Buyer(s) Initials | Seller(s) Initials          |
|                   |                             |

| •        |              |  |
|----------|--------------|--|
|          | 1083         | ADDITIONAL PROVIDING (The fellowing different and the fell |
|          | 1084<br>1085 | , , , , , , , , , , , , , , , , , , ,  |
|          | 1086         | riour Ediato Gorinniosion.)  |
|          | 1087         | 1. SELLER REPRESENTS THE PROPERTY IS NOT SUBJECT TO ANY MORATORIUM THAT WOULD  |
|          | 1088         | PRECLUDE BUYER FROM BUILDING A SINGLE FAMILY RESIDENTIAL HOME.   |
|          | 1089<br>1090 |  |
|          | 1091         | 2. SELLER REPRESENTS THAT TO ITS KNOWLEDGE THERE ARE NO ADVERSE ENVIRONMENTAL  |
|          | 1092         |  |
|          | 1093         | USE FOR THE CONSTRUCTION OF A SINGLE FAMILY RESIDENCE.   |
|          | 1094<br>1095 |  |
|          | 1095         | 3. BUYER REPRESENTS HE IS A COLORADO LICENSED REAL ESTATE AGENT AND MANAGER OF   |
|          | 1097         |  |
|          | 1098         | EFFECTUATE THE CONSUMMATION OF THIS TRANSACTION.   |
|          | 1099<br>1100 |  |
|          | 1101         | 4. IT IS FURTHER AGREED BY THE PARTIES THAT MEC, MUTUAL EXECUTION OF CONTRACT,   |
|          | 1102         | BEGINS FOLLOWING CITY COUNCIL APPROVAL OF THIS AGREEMENT.  |
|          | 1103         | DEGING FOLLOWING CITY COUNCIL AFFROVAL OF THIS AGREEMENT.  |
|          | 1104         | E. Duyar antenny ladges that the Present, year he appared into and achiest to the applicable   |
|          | 1106         | 5. Buyer acknowledges that the Property may be annexed into and subject to the applicable  |
|          | 1107         | Declaration of Covenants, Conditions and Restrictions of Canyon View, as are more particularly   |
|          | 1108         |  |
|          | 1109<br>1110 | Colorado Cicir ana riccorder a Cirico.   |
|          | 1111         | 31. ATTACHMENTS.   |
|          | 1112         | 31.1. The following attachments are a part of this Contract: n/a   |
|          | 1113<br>1114 |  |
|          | 1115         | 31.2. The following disclosure forms are attached but are not a part of this Contract: n/a   |
|          | 1116         | ·  |
|          | 1117         | SIGNATURES   |
|          | 1118<br>1119 |  |
|          | 1120         | Date: 2/19/13  |
|          | 1121         | Buyer Iron Key Homes, LLC  |
|          | 1122<br>1123 | By: Douglas Ferraro, Managing Partner  |
|          | 1124         | Address: 16549 W 15TH AVE Golden CO 80401  |
|          | 1125<br>1126 | Phone: 3038887436 Fax: 3032735694  |
|          | 1127         | Electronic Address: doug@ironkeyhomes.com  |
|          | 1128         |  |
|          | 1129<br>1130 |  |
|          | 1131         |  |
|          | 1132         | [NOTE: If this offer is being countered or rejected, do not sign this document. Refer to §32]  |
|          | 1133<br>1134 | $\mathcal{M}$  |
|          | 1135         |  |
|          | 1136         | Date: 2 AR   |
|          | 1137<br>1138 | Seller: City of Golden   |
|          | 1139         | By: City of Golden   |
|          | 1140         | Address:911 10th Street Golden CO 80401  |
|          | 1141<br>1142 | Phone: 303.384.8097 Fax:   |
|          | 1143         | Electronic Address:  |
|          | 1144         |  |
|          | 1145<br>1146 | ·  |
|          | 1147         | 32. COUNTER; REJECTION. This offer is Countered Rejected.  |
|          | 1148         | Initials only of party (Buyer or Seller) who countered or rejected offer   |
|          | 1149         | 0.14   |
|          | 1150<br>1151 | City of Golden   |
|          | 1152         | By: City of Golden   |
|          | 1153         | END OF CONTRACT TO BUY AND SELL REAL ESTATE  |
|          | 1154<br>1155 |  |
|          | 1156         |  |
|          | 1157<br>1158 | 33. BROKER'S ACKNOWLEDGMENTS AND COMPENSATION DISCLOSURE.  |
|          | 1158         | (To be completed by Broker working with Buyer)   |
|          | 1160         | (1.4.2.2.4.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.  |
|          | 1161<br>1162 | Broker Deep Not seknowledge receipt of Formest Manay deposit and while not a markets the   |
|          | 1163         | Broker Does Does Not acknowledge receipt of Earnest Money deposit and, while not a party to the  |
|          | 1164         | Contract, agrees to cooperate upon request with any mediation concluded under § 23. Broker agrees that if  |
|          | 1165<br>1166 | Brokerage Firm is the Earnest Money Holder and, except as provided in § 24, if the Earnest Money has not   |
| CBS4-9-1 |              | already been returned following receipt of a Notice to Terminate or other written notice of termination, Earnest   |
| CD34-9-  | 12. (        | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 14 of 15  |

Buyer(s) Initials

Seller(s) Initials

| •        |                                      |   |  |
|----------|--------------------------------------|---|--|
|          | 1167<br>1168<br>1169<br>1170<br>1171 | Earnest Money shall be made within five days of Earnest Money Holder's receipt of the executed written mutual instructions, provided the Earnest Money check has cleared.   |  |
|          | 1172<br>1173                         |   |  |
|          |                                      | Brokerage Firm's compensation or commission is to be paid by   Listing Brokerage Firm  Buyer  Other NONE - NO COMMISSION DUE  |  |
|          |                                      | Brokerage Eirm's Name: Colorado Land Professionals, LLC  Date: 2/19/13  Broker's Name: Douglas Ferraro  |  |
|          |                                      | Address: 1505 Quaker Street Golden, CO 80401 Ph: 303.888.7436 Fax: 303.273.5694 Email: coloradolandpros@gmail.com   |  |
|          |                                      | 34. BROKER'S ACKNOWLEDGMENTS AND COMPENSATION DISCLOSURE. (To be completed by Broker working with Seller)   |  |
|          |                                      | Broker Does Does Not acknowledge receipt of Earnest Money deposit specified in § 4.1 and, while not a party to the Contract, agrees to cooperate upon request with any mediation concluded under § 23. Broker agrees that if Brokerage Firm is the Earnest Money Holder and, except as provided in § 24, if the Earnest Money has not already been returned following receipt of a Notice to Terminate or other written notice of termination, Earnest Money Holder shall release the Earnest Money as directed by the written mutual instructions. Such release of Earnest Money shall be made within five days of Earnest Money Holder's receipt of the executed written mutual instructions, provided the Earnest Money check has cleared. |  |
|          |                                      | Broker is working with the Seller as a $\square$ Seller's Agent $\square$ Buyer's Agent $\square$ Transaction-Broker in this transaction. $\square$ This is a Change of Status.   |  |
|          |                                      | Brokerage Firm's compensation or commission is to be paid by $\square$ Seller $\square$ Buyer $\square$ Other $n/a$ Brokerage Firm's Name: $n/a$  |  |
|          |                                      | Broker Date:  |  |
|          |                                      | Address: n/a n/a, n/a n/a   |  |
|          |                                      | Ph: n/a Fax: n/a Email:  CBS4-9-12. CONTRACT TO BUY AND SELL REAL ESTATE (LAND)   |  |
|          |                                      | CTM eContracts - ©2012 CTM Software Corp.   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
|          |                                      |   |  |
| CBS4-9-1 | 12. C                                | CONTRACT TO BUY AND SELL REAL ESTATE - Land Page 15 of 15   |  |
|          |                                      | $\mathcal{I}$   |  |
|          | ) Initia                             | als Seller(s) Initials  |  |

# **Canyon View Homeowners Association**

December 12, 2012

TO: City of Golden

911 10th Street Golden, CO 80401

ATTN:

Marjorie Sloan, Mayor

Marcie Miller, Councilwoman, District 2 Bob Vermeulin, Councilman, Ward 3

Steve Glueck, Director Planning & Development

SUBJECT: Canyon View Lot #1 and Norman D. Memorial Park

RE:

1. City of Golden, Department of Planning & Development Letter, April 4, 2012

2. Canyon View Letter response, August 31, 2012

Dear Sir or Madam:

This letter is to reaffirm our position regarding the City's proposal to sell Lot #1 within our community and to use the proceeds for future improvements to Norman D. Memorial Park.

During the past few weeks, we have surveyed all homeowners in the Canyon View community. Homeowners responded with a clear majority participation rate of 72% and a super-majority of respondents (82%) favored the sale of the City owned Lot #1 (826 Joseph Circle) with net proceeds going toward improvements to Norman D. Memorial Park. Based on the predominate opinion of our homeowners, the Canyon View Board of Directors endorses passage of City Ordinance No. 1925 without further delay.

Sincerely,

Canvon View HOA

Jill Willey, President - jetwil0709@comcast.net, 864-360-1241

Sarah Adler, Vice President

Jon Rieger, Secretary/Treasurer

Bob Nettleship, DRC Liaison

Rob Owens, Member at Large

# **Canyon View Homeowners Association**

August 31, 2012

TO: City of Golden

911 10th Street Golden, CO 80401

ATTN:

Marjorie Sloan, Mayor

Marcie Miller, Councilwoman, District 2 Bob Vermeulin, Councilman, Ward 3

Steve Glueck, Director Planning & Development

SUBJECT: Canyon View Lot #1 and Norman D. Memorial Park

RE: City of Golden, Department of Planning & Development Letter, April 4, 2012

Dear Sir or Madam:

This letter is to express our position regarding the City owned lot in our development (Lot #1) and our plans and expectations regarding future improvements to Norman D. Memorial Park which is adjacent to our neighborhood.

In the referenced letter from Steve Glueck, we were asked for our comments regarding the possibility of Lot #1 being sold for the construction of a single family dwelling. At that time we replied that we did not believe this was in the best interests of our community for a variety of reasons. However, it is our understanding that this issue has been re-formulated and that the City may consider selling Lot#1 and use the proceeds to improve and expand the offerings of Norman D. Park. With this possibility in mind we would like to state our support for this initiative. However, our first choice would be to have the City improve and maintain Lot#1.

If the City chooses not to improve and maintain Lot#1, the Canyon View Homeowners Association strongly supports the sale of Lot#1 for the purpose of constructing a single family dwelling if the following three provisions are met:

- 1. That there is a specific contractual requirement that the Canyon View HOA retain control of the architectural style, design, and detail of the dwelling, as well as approval of any and all landscape design and construction.
- 2. That there is a contractual requirement for the dwelling and homeowner to become part of the Canyon View Homeowners Association.
- 3. That any and all proceeds from the sale of the property (Lot#1) be used for the improvement of Norman D. Memorial Park.

If the City chooses to sell Lot#1, we believe the best possible outcome for the City as well as Canyon View and other nearby communities such as the North Forty and Mesa Meadows includes: (1) the City would divest of unused real estate as well as the associated long-term maintenance costs, (2) the City would receive real estate tax income from the property, (3) funds would be available to improve Norman D. Memorial Park -- a high priority for our community as well as surrounding neighborhoods. Removing Norman D. Park from the current funding priority list would also free up future funding for other City park development requirements. This strategy appears to be a positive solution with no negative impacts.

As we have indicated above, improvements to Norman D. Memorial Park are a very high priority for our community. Canyon view was completed earlier this year and now consists of 59 homes and almost 50 children under the age of 8 years old. Our neighborhood has direct walking access to the park and park improvements are urgently needed to meet current and future recreational needs of the surrounding community. We have been informed that our sister neighborhood of Mesa Meadows is in full support of this initiative also.

In summary, we ask your support of our position. The sale of Lot#1, with the noted conditions would be a "win-win" solution for all parties involved.

Please rely to the email below with any comment or question, and confirm the date/time of the September City meeting when this matter will be discussed.

Sincerely,

Canyon View HOA

Jill Willey, President – <u>jetwil0709@comcast.net</u>, 864-360-1241

Bob Nettleship, Vice President

Jon Rieger, Secretary

Deborah Dauenheimer, DRC Liaison

Blaze Cook, Member at Large